# STUDIES ON THE PROPAGATION OF ERRORS IN PHYSICAL OCEANOGRAPHIC COMPUTATIONS

THESIS SUBMITTED TO THE UNIVERSITY OF COCHIN IN PARTIAL FULFILMENT OF THE REQUIREMENTS FOR THE DEGREE OF

> DOCTOR OF PHILOSOPHY IN PHYSICAL OCEANOGRAPHY

> > Вy

K. S. NEELAKANDAN NAMBOODIRIPAD, M. Sc.

DEPARTMENT OF MARINE SCIENCES, UNIVERSITY OF COCHIN COCHIN-682016

MARCH-1984

### CERTIFICATE

This is to certify that this thesis bound herewith is an authentic record of the research carried out by Mr. K.S.Neelakanden Namboodiripad, M.Jc., under my supervision and guidance in the Department of Marine Sciences, in partial fulfilment of the requirements of the Ph.D. Degree of the University of Cochin and no part thereof has been presented before for any other degree in any University.

Cochin - 682 016, March, 1984.

Dr.P.G.Kurup (Supervising Teacher)



### DECLAR.TIO.F

I hereby declare that the thesis entitled, 'STUDIES ON THE PROPOGRIEGA OF ERRORS IN PROSICAL OCEANOGRAPHIC COMPUTATIONS', is an authentic record of research carried out by me under the supervision and guidance of Dr.P.G.Kurup in partial fulfilment of the requirements of the Ph.D. Degree of the University of Cochin and that no part of it has previously formed the basis for the award of any degree, diploma or associateship in any University.

Dept. of Marine Sciences University of Cochin Cochin - 682 016.

Mulakan ahorm.

(K.S.NEELAKANDAN NAMBOODIRIP Lecturer

#### ACKNOWLEDGEMENTS

The author wishes to record his deep sense of gratitude to Dr.P.G.Kurup, Reader in Physical Oceanography, Department of Marine Sciences, University of Cochin, for suggesting the research problem and for guidance and constan encouragement. He is also grateful for his valuable advice and help in preparation of the manuscript and for suggesting improvements.

He is grateful to Prof.(Dr.) C.V.Kurian, Emeritus Scientist, former Head of the Department of Marine Sciences for encouragement. He is thankful to his colleagues for help and cooperation. He also likes to record his thanks to the Head of the Department of Marine Sciences and University of Cochin for providing necessary facilities.

The help rendered by Sri K.V.Chandran, Senior Grade Typist/Stenographer, University of Cochin, in typing the thesis is gratefully acknowledged. CONTENTS

C. IAP fER	I	INTRODUCTION	Page 1
	1.1.	Physical oceanographic practices	З
	1.2.	Oceanographic computations	9
	1.3.	Scheme of present work	17
CHAPTER	II	THEORY OF ERROAS	19
	2.1.	Errors of observations and measurements	19
	2.2.	Errors in approximations	20
	2.3.	Errors in computations	21
	2.4.	The law of propogation of errors	33
CHAPTER	III	ERRORS IN THE HEASUREMENT OF THE INDEPENDENT OCENHOURAPHIC PARAMETERS	34
	3.1.	Temperature	34
	3.2.	Pressure	40
	3.3.	Salinity	48
	3.4.	On the accuracy of <u>in situ</u> salinometers	59
CHAPTER	VI	ERRORS IN THE COMPUTATION OF DEPENDENT OCE. NOGRAPHIC V. KTABLES	63
	4.1.	Jensity in situ and sigma-t	<b>63</b>
	4.2.	Specific volume and specific volume anomaly	72

## Page

CHAPT ER	J	ERRORS IN THE COMPUTATION OF DERIVED QUANTITIES	77
	5.1.	Dynamic depth anomaly of isobaric surfaces	77
	5.2.	Relative currents	88
CHAP (ER	VI	IDENTIFICATION OF A ZEAU REFERENCE LEVEL	94
	5.1.	Determination of an appropriate reference surface	94
	6.2.	Identification of a level where the current velocity is negligibly small	102
CHAP FER	VII	COMPUTATION OF JIMAMIC HEIGHT Anomaly	114
	7.1.	Extending the selected zero reference surface into shallow regions	115
	7.2.	A new method suggested for the extension of computation into shallow regions	120
	7.3.	Smoothening the dynamic celief in a profile of an isobaric surface	123
CHAPTER	VIII	CUMPUTATION OF GEOSTRUP IIC CURRENTS	126
	8.1.	Reduction of the computed geostrophic current with reference to 1500 d bar surface to 3000 d bar surface	- 126
	3.2.	Summary and conclusions	134
REFERENCES			143

### CHAPTER I

### INTRODUCTION

Systematic procedure of physical oceanographic research involves collection of data on the independent parameters of the physical properties of sea water and the computation of the dependent parameters, leading ultimately to information on the dynamics of the oceans. The data collected at sea include, among others, values of temperature and salinity at surface and subsurface levels. Physical oceanographic measurements made at sea are subjected to careful examinations, corrections and conversions which require considerable amount of practical experience to judge the reliability of the records from instruments which operate blindly below the sea surface. Corrections are made for instrumental errors and for errors inherent in the methods of obtaining the data. The oceanographer is thus equipped with the basic data of temperature and salinity for different depths at various stations in the sea. The processing procedures do not ultimately provide perfectly accurate information on these parameters including temperature, salinity, depth, station position, etc. These procedures provide the data within certain error limits.

The basic data are then converted to standard depths to facilitate comparison with other oceanographic

data. This is followed by a series of calculations required to derive the dependent quantities like specific volume, density and currents. These calculations have been highly systematised through practice by the physical oceanographers. This branch of physical oceanography is known as dynamical oceanography. Dynamical oceanography discusses the water properties and water movements and their temporal and spatial variations in the world oceans.

Dynamical oceanography has almost always neglected to consider as to what happens to the random errors, inherent in the basic data in the process of the series of calculations involved in the computational practices. Ιt is conveniently assumed that the final results are not much affected by these errors. The fact that even small errors can, at times, lead to highly euraneous results through the propagation of errors in the computational procedures is often overlooked by oceanographers. Very few authors have made any serious mention on the error component of the derived results in dynamical oceanography. Jakhelln (1936), Thompson (1939) and Dobrovol'skii (1949) have mentioned this possibility, while Fomin (1964) has given serious consideration to this problem. The present studies aim to examine the limits of ecrors contained in

-2-

the basic data of physical oceanography and the magnitude of the error component in the results derived in dynamical oceanography. The study also suggests a graphical method of smoothening of the derived parameters within the limits of errors to increase the reliability of the derived results.

### 1.1. Physical Oceanographic Practices

Customerily and scientifically, the practical procedures in Physical Oceanography include collection of the various data during oceanographic cruises. Of these, those that are relevant in the context of errors and their propagation will be discussed in this section.

### 1.1.1. Station position

The place where an oceanographic vessel is stopped to carry out observations and collection of data and samples is known as an oceanographic station. The geographic location of the station is a primary requirement in oceanography. Then land is visible an if recognisable features are accurately located on land, the position of the vessel is determined by means of horizontal angles and bearings on shore features. Out of sight of land,

-3-

the ship's position is determined by astronomic sights of by radio direction-finder bearings. Between positions established in these ways, the location at any time is obtained by dead-reckoning, from the course steered and distance run. More recently, satellite navigation systems are available which provide more accurate information on the location of oceanographic stations. The values of latitudes are of particular interest as it appears in the dynamic computation as a term in the Corfolis force component. The station positions also betermine the values of distance between stations which again is a term in the dynamic computation for estimating relative currents.

### 1.1.2. Depth

Depth of the bottom at the station is obtained in shallow regions by the use of the classical method of lead sounding. More accurate and easy method is provided by echosounding in which the time interval between the generation of a sound impulse and the reception of its echo is used as a direct measure of the depth, using a constant sounding velocity. Depth of the oceanographic station does not directly appear in the uphanic computations and therefore does not require detailed considerations in the present study.

-4-

### 1.1.3. Temperature

Measurement of surface temperature is carried out using accurate mercury-in-glass thermometers. Thermometers used for measuring temperatures at subsurface levels are of the reversing type and are generally mounted upon water sampling bottles so that temperature and the water for salinity and other physical and chemical tests are obtained at the same level. Serial data on temperature and salinity ace thus obtained from a hydrographic cast of water samplers arranged in series on a wire rope, to each of which are attached the protected and unprotected reversing thermometers.

The protected reversing thermometer is essentially a double ended thermometer. It is sent down to the require depth in the set position and consists of a large reservoir of mercury connected by means of a fine capillary to a small bulb at the upper end. Just above the large reservoir, the capillary is constricted and branched with a small arm, and above this, the thermometer tube is pent into a loop, from which it continues straight and terminates in the smaller bulb. In the set position, mercury fills the reservoir, the capillary and part of the bulb. The amount of mercury above the constriction

-5-

depends upon the temperature of the surrounding water. When the thermometer is reversed at the required depth along with the water bottle by sending a messenger weight down the wire rope, the mercury column breaks at the point of constriction and runs down, filling the bulb and part of the graduated capillary, thus indicating the temperature at reversal. The loop in the capillary which is generally of enlarged diameter, is designed to trap any mercury that is forced past the constriction if the temperature is raised after the thermometer has been reversed. In order to correct the reading for the changes resulting from difference between the temperature at reversal and surrounding temperature at the time of reading, a small standard thermometer, known as the auxiliary thermometer, is mounted alongside the reversing thermometer. fhe reversing thermometer and the auxiliary thermometer are enclosed in a heavy glass tube that is partially evacuated except for the portion surrounding the reservoir of the thermometer, and this part is filled with mercury to serve as a thermal conductor between the surroundings and the reservoir. The thermometer tube eliminates the effect of hydrostatic pressure. Seawater temperature in situ is obtained from the reading of a protected reversing

-6-

thermometer by applying corrections for instrumental error and for thermal expansion subsequent to reversal.

Reversing thermometers were first introduced by Negretti and Zambra in 1874 and have since been improved so that well made instruments are now accurate to within  $\pm 0.01^{\circ}$ C. Recently electronic instruments are also being used to obtain records of subsurface temperature.

### 1.1.4. Depth of sampling

By depth of sampling we mean the subsurface depth at which the reversing bottle along with the reversing thermometer is made to reverse collecting subsurface water sample and recording the temperature. The wire rope carrying the equipment is payed out through a meter wheel which measures the length of wire rope that has run out. The depth of reversal, as obtained from the meter wheel readings, may be erraneous due to non-vertical running out of the wire rope in the presence of ship drift or ocean currents. A correction for this can be obtained in the surface layers by measuring the wire angle. Depth of reversal is more accurately found by comparing the corrected reading of protected thermometer with the corrected reading of unprotected thermometer which is paired with a protected thermometer.

. -7-

Unprotected reversing thermometers are identical with the protected reversing thermometers but have open protective tubes. Because of the difference in the compressibility of glass and mercury, thermometers subjected to pressure give a fictitious temperature reading that is dependent upon the temperature and pressure. The unprotected reversing thermometers are so designed that the apparent temperature increase due to hydrostatic pressure is about  $0.01^{\circ}$ C/m. The readings obtained from the unprotected thermometers also have to be corrected for thermal expansion and instrumental errors.

### 1.1.5. Salinity

Water samples for estimation of salinity are obtained in oceanographic vessels using subsurface cast of Nansen bottles in series. The Nansen bottle is a reversing bottle which can be reversed at the desired depth by sending a messenger weight down the wire rope. On reversal the bottle closes entrapping the water at that depth.

Salinity is directly proportional to chlorinity, which is determined by the titration of the water sample with silver nitrate in the presence of a suitable indicator.

6 |

Recently electrical conductivity is used as a measure of salinity and STD or CTD recorders are available for speedy collection of hydrographic data.

### 1.2. Oceanographic computations

There are a series of computations which the oceanographers do to derive the values of the various dependent quantities from the hydrographic data. These operations begin with the interpolation of the values of temperature and salinity at standard depths and conclude with the computation of currents from the distribution of density as obtained from the temperature and salinity data.

### 1.2.1. Conversion to standard depths

The first step in the processing of serial data on temperature and salinity is to prepare plots for the vertical distribution of the variables. Such plots of temperature and salinity as a function of depth are useful to detect incorrect values resulting from faulty operation of thermometers and water bottles. Another use of these plots is to scale off depths of decided values of the variables which are necessary for drawing diagrams of horizontal distributions. The main purpose of plotting

-9-

variables against depth is to obtain interpolated values of temperature and salinity at 'Standard depths'. The International Association of Physical Oceanography has defined standard depths as: Surface, 10,20,30,50,75,100, 150,200,(250),300,400,500,600,(700),800,1000,1200,1500, 2000,2500,3000 and 4000 metres and intervals of 1000 metres thereafter to the greatest depth of sampling. The National Oceanographic Data Centre has accepted the following standard depths viz., Surface, 10,20,30,50,75,100,125,150, 200,250,300,400,500,600,700,800,900,1000,1100,1200,1300, 1400,1500,1750,2000,2500,3000,4000 metres and intervals of 1000 m to the greatest depth of sampling, which are currently being used.

Femperature and salinity data are also plotted on a T-S diagram. Introduced by Helland-Hansen (1916), the temperature-salinity diagrams are polled with salinity on the X-axis and temperature on the y-axis. When salinities are plotted against temperatures, the points generally lie on a well-defined curve. The T-S curve gives the temperature-salinity relationship of the sub-surface water in the area under study. Surface data have to be omitted because annual variations and local

-10-

modifications lead to descrepancies. The I-S diagram helps to detect errors and to bring out watermass characteristics of the data. After the T-S curve has been drawn for the observed data, corresponding interpolated values of temperatures and salinity read from the vertical curves are also plotted. If these data do not fall on the T-S curve, certain adjustments must be made in the construction of the vertical distribution curves.

# 1.2.2. Computation of specific volume and specifiv volume anomaly

After obtaining the values of depth, temperature and salinity as detailed above, certain calculations are necessary to derive the values of the various dependent variables commonly used to describe the field of mass in the sea. These variables include: specific volume, anomaly of specific volume from a standard value, density <u>in situ</u> and  $\sigma_+$ , which represents density at surface pressure.

Specific volume is volume per unit mass. Specific volume in situ in the sea is expressed by the symbol  $\alpha_{s,t,p}$  where the subscripts indicate salinity, temperature and pressure of the sample. Specific volume is computed by expressing it as a known specific volume under given conditions plus a series of correction terms for the

-11-

dependent variables of temperature, salinity and pressure. These terms may be grouped, computed, and added as follows to give specific volume <u>in situ</u>.

 $\alpha_{s,t,p} = (\alpha_{35,0,0} + \delta_p) + (\delta_s + \delta_t + \delta_{s,t}) + \delta_{p,p} + \delta_{t,p} + \delta_{s,t,p}$ In the first term  $\alpha_{35,0,0}$  is a constant (0.97264) and  $\delta_p$  represents the effect of pressure at standard salinity (35%.) and at 0°C.

The next term depends only on salinity and temperature and may be summed to give  $\Delta_{s,t}$  or  $\delta_T$ , known as the thermosteric anomaly. This term is found from values of temperature and salinity by means of tables or graphs (Sverdrup, 1933). The salinity-pressure term,  $\delta_{s,p}$  and the temperature-pressure term  $\delta_{t,p}$  are also found from tables or graphs. The last term,  $\delta_{s,t,p}$ , is so small that it is always neglected.

The sum of the terms  $\Delta_{s,t}$ ,  $\delta_{s,p}$  and  $\delta_{t,p}$ constitutes the anomaly of specific volume from the standard  $\alpha_{35,0,p}$  and is designated by the symbol  $\delta$ . For computing currents, the variations in specific volume along isobaric surfaces are required. Since pressure is constant along any given isobaric surface, the term  $\delta_p$  is a constant. It is sufficient therefore to calculate the specific volume anomaly,  $\delta$ , since the standard term does not contribute to variation in specific volume along an isobaric surface.

Graphs or tables are available for the calculation of specific volume anomaly (Sund, 1926; LaFond, 1940; Callaway, 1950). LaFond (1951) presents the oceanographic tables for finding out the values of the various terms, the sum of which gives the specific volume anomaly.

### 1.2.3. Computation of density and sigma-t

Density in situ is the reciprocal of specific volume in situ. Another way to express the density is by the symbol  $\sigma_{s,t,p}$ . By definition  $\sigma_{s,t,p}$  is equal to  $10^3$  ( $P_{s,t,p}$ -1). This expression has the advantage that the numerical value contains fewer digits and is easier to handle.

 $\sigma_t$  represents the density of water of given salinity and temperature at surface pressure. In oceanography,  $\sigma_t$  assumes significance because the motion along  $\sigma_t$  surfaces involves little change in energy and therefore mixing of water masses tends to take place along these surfaces. After calculation  $\Delta_{s,t}$ ,  $\sigma_t$  is obtained directly from the table for  $\sigma_t$  for values of  $10^5 \Delta_{s,t}$  (Sverdrup, 1933). Based on Knudsen's equations (Knudsen, 1901), several authors have compiled tables for  $\sigma_t$  for values of temperature and salinity (Mc Ewen, 1929; Mathews, 1932; Fleming, 1939; Ennis, 1944; Bumpus and Martineau, 1948).

### 1.2.4. Computation of currents

Dynamic computations provide information on relative currents pertaining to the distribution of mass in the sea. Such relative currents are deduced from a consideration of the balance of forces in the sea. The forces considered are those which act along an isobaric surface. When the isobaric surface is not level, a component of gravity acts downward along it. This is balanced by the Coriolis force so that the slope of the surface is maintained.

The thickness of an isobaric layer, which is the layer between two isobaric surfaces, depends upon the average specific volume of the layer. Therefore, the slope of an isobaric surface relative to another, which is assumed to be level, may be found. Since the dynamic height is a measure of the work performed against gravity in moving unit mass from one level to another, the component of the force of gravity acting down the slopping isobaric surface between two stations is the difference in dynamic height of the surface at the two stations divided by the distance between the stations. Equating This expression to the expression for Coriolis force, we can obtain the component of current negatal to the line joining the two stations (Sandstrom and Helland-Hansen, 1903). This current is at the upper isobaric surface and is relative to any current which is present at the lower reference surface. In dynamic computation, the isobaric surface is assumed level at some depth where the motion is negligible and the dynamic slope of an upper isobaric surface is found from the variation of specific volume along the isobaric layer. Thus, the current at the upper surface relative to any possible current at the lower surface is determined.

For each oceanographic station, the dynamic thickness of the isobaric layer is calculated by means of the equation,

$$D_2 - D_1 = \int_{p_1}^{p_2} \alpha \, dp$$

where  $D_2 - D_1$  is the dynamic thickness of the isobaric layer,  $\alpha$  is the specific volume and dp is the pressure

-15-

interval. Since  $\alpha = \alpha_{35,0,p} + \delta$ , the total dynamic thickness of the layer may be considered as the sum of the dynamic thickness of the layer of standard specific volume and the increment in dynamic thickness due to the anomaly of specific volume from the standard. Since the dynamic thickness of the standard layer is the same at every station, the differences in dynamic height between stations are given by the differences in the increments which can be obtained as  $\int_{-2}^{-2} \delta dp$ . In this, metres of depth are substituted for decibars of pressure.

The dynamic computation involves the following procedures. The specific volume anomaly,  $\delta$ , at each depth is calculated. The mean specific volume anomaly,  $\delta$ , for each depth interval is determined by averaging the two bounding values. This is multiplied by the depth interval to get the anomaly of the dynamic height,  $\Delta D$ , for each small depth layer. Fotal  $\Delta D$  for each station is obtained by adding the anomalies of dynamic height from the selected reference level to the level at which relative currents are to be computed. The relative current velocity normal to a line joining two stations is obtained in metres/second as

$$V = \frac{10(\Delta D_A - \Delta D_B)}{L 2 \omega \sin \phi}$$

where  $\Delta D_A - \Delta D_B$  is equal to the difference in the anomalies in the dynamic height at stations A and B in dynamic metres, L is the distance between the stations in metres,  $\omega$  is the angular velocity of the earth (0.789 x 10<sup>-4</sup> radians/sec.) and  $\phi$  is the mean latitude between the stations.

### 1.3. Scheme of the present work.

The results of an investigation on the limits of the random errors contained in the basic data of Physical Oceanography and their propagation through the computational procedures are presented in this thesis. It also suggest a method which increases the reliability of the derived results. The thesis is presented in eight chapters including the introductory chapter. Chapter 2 discusses the general theory of errors that are relevant in the context of the propagation of errors in Physical Oceanographic computations. The error components contained in the independent oceanographic variables namely, temperature, salinity and depth are deliniated and quantified in chapter 3. Chapter 4 discusses and derives

the magnitude of errors in the computation of the dependent oceanographic variables, density in situ,  $\sigma_{+},$  specific volume and specific volume anomaly, due to the propagation of errors contained in the independent oceanographic variables. The errors propagated into the computed values of the derived quantities namely, dynamic depth and relative currents, have been estimated and presented chapter 5. Chapter 6 reviews the existing methods for the identification of level of no motion and suggests a method for the identification of a reliable zero reference level. Chapter 7 discusses the available methods for the extension of the zero reference level into shallow regions of the oceans and suggests a new method which is more reliable. A procedure of graphical smoothening of dynamic topographies between the error limits to provide more reliable results is also suggested in this chapter. Chapter 8 deals with the computation of the geostrophic current from these smoothened values of dynamic heights, with reference to the selected zeroreference level. The summary and conclusion are also presented in this chapter.

-18-

### CHAPTER II

### THEORY OF ERRORS

The results of experimental observations will have always inaccuracies. It is necessary to know the magnitude of these inaccuracies in order to arrive at the inaccuracy in a computed result using the observations. A knowledge of the statistical behaviour of the inaccuracies or errors of observations will be of great help in reducing the effect of those uncertainties in the final result.

### 2.1. Errors of observations and measurements

All observations and measurements are subject to three kinds of errors: systematic errors, accidental or random errors, and mistakes.

Systematic errors are those which affect all measurements alike. They are mostly due to imperfections in the construction or adjustment of the instrument, the 'personal equation' of the observer, etc. Such errors can be remedied by applying proper corrections.

Accidental or random errors are those whose causes are unknown and indeterminate. They are usually small.

It is found experically that such random errors are frequently distributed according to a simple law, the law of chance. This makes it possible to use statistical methods to deal with random errors. The mathematical theory of errors deals with random errors only.

Mistakes are not, properly speaking, errors at all. They are blunders performed during reading of an instrument, recording of a result and in computations. They can be eliminated by careful work.

The word 'precision' is used in relation to random errors. A precise measurement will be free from random errors. An accurate measurement is one that is free from all kinds of errors - systematic, random and mistakes.

### 2.2. Errors in approximations

Numbers such as 2, 1/3, etc. are known as exact numbers because there is no uncertainty in them. On the other hand, numbers such as  $\pi$ ,  $\sqrt{3}$ , etc. even though exact, cannot be expressed exactly by a finite number of digits. These numbers, when expressed in digital form, are known as approximate numbers. Aumbers of the above type and quotients of division which never terminate will have to be out down to a manageable size to be used in practical computations. This process of cutting off superfluous digits and retaining as many as desired is known as 'rounding off'. This process, obviously, introduces an error into the number. The following rule for rounding off will cause the least possible error.

'To round off a number to n significant figures, discard all digits to the right of the n<sup>th</sup> place. If the discarded number is less than half a unit in the n<sup>th</sup> place, leave the n<sup>th</sup> digit unchanged; if the discarded number is greater than half a unit in the n<sup>th</sup> place, add one to the n<sup>th</sup> digit. If the discarded number is exactly half a unit in the n<sup>th</sup> place, round off so at to leave the n<sup>th</sup> digit an even number' (Scarborough, 1965). When a number is rounded off according to this rule, it will be correct to n significant figures. The error introduced into the number due to rounding off will not be greater than half a unit in the n<sup>th</sup> significant figure.

### 2.3. Errors in computations

When various quantities are used to calculate a result and when the different quantities have errors in

-21-

them, then the result will also be in error by an amount which depends on the errors of the individual quantities.

### 2.3.1. The general formula for errors

Let .  

$$Q = f(a_1, a_2, a_3, \dots, a_n)$$
 (2.1)

be a function of several independent quantities  $a_1, a_2, \ldots, a_n$  which are subject to errors  $da_1, da_2, \ldots, da_n$ . These errors in a's will cause an error, dQ, in the function Q according to the relation.

$$Q + \partial Q = f (a_1 + \partial a_1, a_2 + \partial a_2, \dots, a_n + \partial a_n)$$
 (2.2)

To find an expression for dQ, we must expand the right hand side by Taylor's theorem.

Since the errors  $\mathbf{d}_{a_1}, \mathbf{d}_{a_2}, \ldots$  are relatively small we may neglect their squares, products and higher powers and so

$$\mathbf{Q} + \mathbf{d}\mathbf{Q} = \mathbf{f}(\mathbf{a}_1, \mathbf{a}_2, \dots, \mathbf{a}_n) + \mathbf{d}\mathbf{a}_1 - \frac{\partial \mathbf{f}}{\partial \mathbf{a}_1} + \mathbf{d}\mathbf{a}_2 - \frac{\partial \mathbf{f}}{\partial \mathbf{a}_2} + \dots + \mathbf{d}\mathbf{a}_n \frac{\partial \mathbf{f}}{\partial \mathbf{a}_n}$$
(2.4)

Substracting equation (2.1) from equation (2.4),

$$\mathbf{d}_{Q} = \mathbf{d}_{a_{1}} \frac{\partial f}{\partial a_{1}} + \mathbf{d}_{a_{2}} \frac{\partial f}{\partial a_{2}} + \dots + \mathbf{d}_{a_{n}} \frac{\partial f}{\partial a_{n}}$$
$$= \frac{\partial Q}{\partial a_{1}} \mathbf{d}_{a_{1}} + \frac{\partial Q}{\partial a_{2}} \mathbf{d}_{a_{2}} + \dots + \frac{\partial Q}{\partial a_{n}} \mathbf{d}_{a_{n}} \dots (2.5)$$

This is the general formula for computing the error of a function and gives the absolute error. The expression, obviously, is the total differential of the function Q. The relative error in the function Q is the quotient obtained when the absolute error is divided by the true value of the quantity.

$$\frac{\partial}{\partial q} = \frac{\partial q}{\partial a_1} \frac{\partial}{\partial q} + \frac{\partial q}{\partial a_2} \frac{\partial}{\partial q} + \cdots + \frac{\partial q}{\partial a_2} \frac{\partial}{\partial a_2} + \cdots + \frac{\partial q}{\partial a_n} \frac{\partial}{\partial q}$$
(2.6)

# 2.3.2. Application of the general formula for errors in the fundamental operations of Arithmetic

(i) Addition

Let the function  $\mathbb{Q}$  be of the form

$$Q = a_1 + a_2 + \dots + a_n$$
 (2.7)

Applying the general formula for errors to the function Q, the absolute error  $\Xi_a$  is,

$$E_a = d_Q = d_{a_1} + d_{a_2} + \dots + d_{a_n}$$
 (2.3)

Here each of the  $\mathbf{d}_{a_1}$  are just as likely to be positive as negative. In order to be sure of the maximum error in the function Q, we must take all the terms with the positive sign.

Formula (2.8) indicates that when a few numbers of different accuracies are added it would be useless and absurd to retain all the decimal digits in all the numbers because the error in the result, any way, will be greater than half a unit in the last significant figure of the least accurate number. A safe rule for addition will be to retain one more decimal digit in the more accurate numbers than is contained in the least accurate number and round off the result to the decimal digit contained in the least accurate number. The result usually will be uncertain by one unit in the last figure. Retaining one more digit in the more accurate numbers than is contained in the least accurate number eliminates the possibility of the errors due to rounding off the more accurate numbers from affecting the error in the final result.

Since errors can be just as likely to be positive as negative, their algebraic sum will never be large when a large number of approximate numbers are added. And the mean of several approximate numbers can be more accurate than the numbers from which it was obtained because the computation involves addition as well as division by the total number (Scarborough, 1966).

### (ii) Subtraction

Let the function Q be of the form

$$^{\sim}Q = a_1 - a_2$$
 (2.9)

Applying the general formula for errors to the function Q

$$\mathbf{\dot{a}}_{2} = \mathbf{\dot{a}}_{1} - \mathbf{\dot{a}}_{2} \tag{2.10}$$

Since the errors  $da_1$  and  $da_2$  are as likely to be positive

as negative, we should take the sum of the errors to get the maximum error in the function Q. Hence, the absolute error  $E_a$  is

$$E_a = d_0 = d_{a_1} + d_{a_2}$$
 (2.11)

A safe rule for the subtraction between two numbers of unequal accuracy is to round off the more accurate number to the same number of decimal places as the less accurate number and then subtract. The result usually will be in error by one unit in the last figure.

### (iii) <u>Multiplication</u>

\_ Let the function Q be of the form

$$Q = a_1 \times a_2 \times \dots \times a_n$$
 (2.12)

Taking the total differential and dividing by Q, the relative error  ${\rm E}_{\rm p}$  is,

$$E_{r} = \frac{d_{2}}{Q} = \frac{d_{a_{1}}}{a_{1}^{2}} + \frac{d_{a_{2}}}{a_{2}^{2}} + \dots + \frac{d_{a_{n}}}{a_{n}^{2}}$$
(2.13)

Here again, each of the  $d_a_i$  can be as likely to be positive as negative and in order to be sure of the maximum relative error in the function Q, we must take the sum of the relative errors. The accuracy of a product should be investigated by means of relative error. The absolute error, can be found from the relation

$$E_{a} = E_{r} \times Q \tag{2.14}$$

while finding the product of two or more approximate numbers, the sofe rule is to retain one more significant figure in the more accurate factors than that contained in the least accurate factor and to round off the result to as many significant figures as in the least accurate factor (Scarborough, 1966).

(iv) <u>Division</u>

Let the function be of the form

$$Q = \frac{a_1}{a_2} \tag{2.15}$$

faking the total differential of the function and dividing by Q

$$\frac{\mathbf{d}_{2}}{\mathbf{d}_{2}} = \frac{\mathbf{d}_{a_{1}}}{a_{1}} - \frac{\mathbf{d}_{a_{2}}}{a_{2}}$$
(2.16)

The errors  $\mathbf{\dot{d}}_{a_1}$  and  $\mathbf{\dot{d}}_{a_2}$  are as likely to be positive as negative and to get the maximum relative error in the function, we must take the sum of the relative errors. Hence, the relative error  $\mathbf{E}_r$  is

$$E_{r} = \frac{d_{a_{1}}}{Q} = \frac{d_{a_{1}}}{a_{1}} + \frac{d_{a_{2}}}{a_{2}^{2}}$$
(2.17)

The accuracy of quotients also should be investigated using relative error. The safe rule for division is to retain one more significant figure in the more accurate factor than the number of significant figures contained in the less accurate factor and to round off the result to the number of significant figures contained in the less accurate factor (Scarborough, 1966).

### 2.3.3. The normal law of errors

Since an error can be just as likely to be positive as negative, their algebraic sum will never be large in a computation involving a large number of approximate numbers. In such situations, the error in the result should be obtained using the normal law of errors, since it is generally found that the normal distribution or Gauss distribution describes the distribution of random errors (Young, 1962).

A random error may be assumed to be the result of a large number  $\pounds$  of elementary errors, all of equal magnitude  $\epsilon$ , and each equally likely to be positive or negative. Now we assume the following facts to be true for the distribution of random errors.

- ì) Small errors are more frequent than large errors.
- ii) All errors are equally likely to be positive or negative.
- iii) Very large errors do not occur.

Then the probability of occurrence of an error x in the range  $(-\pounds \epsilon)$  to  $(+\pounds \epsilon)$  is given by the probability equation

$$\gamma = P(x) = \frac{h}{\sqrt{\pi}} e^{-h^2 x^2}$$
 (2.18)

where'h is called the measure or index of precision of the distribution represented by the equation and known as the normal error distribution. The measure of precision h and the standard deviation  $\sigma$  of the distribution is related as

$$h = \frac{1}{\sqrt{2}} \sigma$$
 (2.19)

dence the probability equation can be written in terms of  $\sigma$  as:

$$Y = P(x) = -\frac{1}{\sqrt{2\pi}\sigma} e$$
 (2.20)  
earborough, 1965)

(Sca lgn, 1965)

Equation (2.13) shows that zero error has maximum probability of occurrence and the probability decreases as the magnitude of the error increases. The quantity h is called the 'index of precision' because, for larger values of h the probability of occurrence of zero error is larger and also the probability decreases faster as the magnitude of the error increases. This means that small errors are more frequent than large errors which indicates precision of the data.

For the normal error distribution, the probability of an error to fall within 1,2 and 3 standard deviations from the mean, which is zero error, is given in Table I (Young, 1962).

а — — — — — — — — — — — — — — — — — — —	P(x)	∕ ₽(x)			
1	0.683	68.3			
2	0.954	95.4			
3	0.997	99.7			

<u>Table I</u>

If  $M_1$ ,  $M_2$ ,  $M_3$ , ....,  $M_n$  are **n** independent normal error distributions having standard deviations  $\sigma_1$ ,  $\sigma_2$ ,  $\sigma_3$  ...  $\sigma_n$  respectively, then their sum

 $F = M_1 + M_2 + M_3 \dots + M_n$  (2.21)
will also be a normal error distribution whose standard deviation is given by (Scarborough 1966)

$$\sigma_{\rm F} = \sqrt{\sigma_1^2 + \sigma_2^2 + \sigma_3^2 + \dots + \sigma_n^2}$$
(2.22)

The standard deviation of the normal error distribution which results from the summation of S values, each of which may, with equal probability, contain an error equal to one of the values of the finite sequence

$$-le, -(l-1)e, -(l-2)e, \dots, -2e, -e, 0,$$
  
$$e, 2e, \dots, (l-2)e, (l-1)e, le$$

may be obtained as follows. The standard deviation of the above sequence is given by

$$\sigma = \frac{\int S_{-l}^{2} (n e)^{2}}{\int S_{-l}^{2} (n e)^{2}} = \frac{e \int S_{-l}^{2} S_{-l}^{2}}{\int 2^{l} + 1}$$
  
=  $\frac{e \int S_{-l}^{2} (l + 1) (2l + 1)}{\int 6 (2 l + 1)} = \frac{e \int S_{-l}^{2} (l + 1)}{\sqrt{3}}$ 

Assuming  ${\cal L}$  to be large, the above may be approximated to

$$= \epsilon \frac{\sqrt{\frac{S\ell^2}{\sqrt{3}}}}{\sqrt{3}} *$$

$$= \frac{\sqrt{\frac{S\ell}{\sqrt{3}}}}{\sqrt{3}} *$$
(2.24)

The formula for the standard deviation in this form will be useful when only the maximum value of the individual error,  $\mathcal{L}\epsilon$ , is known

σ

\*Formula (2.24) is derived in the same way as was done by Fomin (1964) except for his assumption that each of the S values of the summation contains an error, with equal probability equal to one of the intigers in the range  $-\not L$  to  $\not L$ . With this assumption he got the formula for the index of precision as

$$h = \frac{\sqrt{6}}{2\sqrt{\ell(\ell+1)S}}$$
(2.25)

The formula for the standard deviation can easily be obtained from the above as

$$\sigma = \frac{\sqrt{S\ell(\ell+1)}}{\sqrt{3}}$$
(2.26)

There is a conceptual error in the above assumption that the error can assume only values of the intigers. Formula (2.26) will also result in different magnitudes for the standard deviation for different units used for the error  $\mathcal{L}$ . 2.4. The law of propagation of errors

Consider a quantity Q which is calculated from several observed quantities  $a_1$ ,  $a_2$ ,  $a_3$  .....

$$Q = f(a_1, a_2, a_3 \dots)$$
 (2.27)

Suppose that a<sub>1</sub>,a<sub>2</sub>,a<sub>3</sub>, ..... are all measured N times. Then the law of propagation of errors is expressed by the equation (Young, 1962)

$$\sigma_{Q}^{2} = \left(\frac{\partial Q}{\partial a_{1}}\right)^{2} \sigma_{a_{1}}^{2} + \left(\frac{\partial Q}{\partial a_{2}}\right)^{2} \sigma_{a_{2}}^{2} + \left(\frac{\partial Q}{\partial a_{3}}\right)^{2} \sigma_{a_{3}}^{2} + \dots (2.23)$$
where  $\sigma_{a_{1}}, \sigma_{a_{2}}, \sigma_{a_{3}}, \dots$  are the standard deviations  
of the quantities  $a_{1}, a_{2}, a_{3}, \dots$  Where the  
observations are not repeated, as in the case of field  
sciences like oceanography, meteorology, etc., the  
equation expressing the law of propagation of errors  
reduces to the general formula for errors expressed by the  
equation (2.5).

#### CHAPTER III

## ERRORS IN THE MEASUREMENT OF THE INDEPENDENT OCEANOGRAPHIC PARAMETERS

Measurements of independent oceanographic parameters, namely, temperature, salinity and pressure, are subject to errors. In this chapter the magnitude of such errors are discussed assuming that only random errors are committed in the measurements.

#### 3.1. Temperature

The temperature of sea water at various depths is measured using a protected reversing thermometer (Sverdrup et al. 1942). To get the true in situ temperature, two corrections are to be added algebraically to the main thermometer reading. The first correction is for the relative expansion of mercury and glass subsequent to reversal, since the ambient temperature at which the thermometer is read may be quite different from that at which it was reversed. Schumacher's formula (Schumacher, 1923) is made use of for this purpose which gives the correction as

$$C = \frac{(T'-t) (T'+V_0)}{K} \left[1 + \frac{(T'-t) + (T'+V)}{K}\right] \quad (3.1)$$

where the symbols have the following meanings:

- C expansion correction to be added algebraically to the main thermometer reading, expressed in <sup>O</sup>C.
- t auxiliary thermometer reading in <sup>O</sup>C.
- T'- Main thermometer reading in <sup>C</sup>C.
- Vo- Volume of mercury below <sup>O</sup>C mark of the main thermometer, expressed in <sup>O</sup>C of the main thermometer scale.
- K reciprocal coefficient of the relative expansion of mercury and thermometer glass. This value for most of the modern thermometers in  $6100^{\circ}$ C.

The second correction, namely the index correction, represented by I, is for the errors caused by the irregular cross section of the capillary and for the irregularities of the scale etching. Hence the total correction to be added algebraically to the protected main thermometer reading is

$$\Delta T = C + I \tag{3.2}$$

and the corrected reading of the protected main thermometer,  $T_w$ , is given by

$$T_{W} = T' + \Delta T$$
$$= T' + C + I \qquad (3.3)$$

Since all the different quantities in the expression (3.2) are subject to measurement errors, addition of the expansion correction C and the index correction I to the uncorrected main thermometer reading introduces errors into the corrected main thermometer reading and this can be obtained by taking the total differential of the above expression (3.3)

$$d I_{W} = dT' + dC + dI \qquad (3.4)$$

The error dI introduced by the addition of the index correction is neglected in the present discussion assuming it to be small since modern reversing thermometers have very uniform capillary bore and also extremely uniform scale etching. An expression for dC is obtained by taking the total differential of the expression for C and is given by

$$dC = \left[ -\left\{ \frac{1}{T'-t} + \frac{1}{K+(T'-t)+(T'+V_{0})} \right\} dt + \left\{ \frac{1}{T'-t} + \frac{1}{T'+V_{0}} + \frac{2}{K+(T'-t)+(T'+V_{0})} \right\} dT' + \left\{ \frac{1}{T'+V_{0}} + \frac{1}{K+(T'-t)+(T'+V_{0})} \right\} dV_{0} + \left\{ \frac{1}{T'+V_{0}} + \frac{1}{K+(T'-t)+(T'+V_{0})} - \frac{2}{K} \right\} dV_{0} - \left\{ \frac{1}{K+(T'-t)+(T'+V_{0})} - \frac{2}{K} \right\} dK \right] C$$

-36-

Now, since each of the errors t, T',  $V_0$ , and  $\kappa$  are just as likely to be positive as negative, we should take all the terms with a positive sign to be sure of the maximum error in C. The final expression for C is:

$$dC = \left[ \left\{ \frac{1}{T'-t} + \frac{1}{K+(T'-t)+(T'+V_0)} \right\} dt + \left\{ \frac{1}{T'-t} + \frac{1}{T'+V_0} + \frac{2}{K+(T'-t)+(T'+V_0)} \right\} dt + \left\{ \frac{1}{T'+V_0} + \frac{1}{K+(T'-t)+(T'+V_0)} \right\} dV_0 + \left\{ \frac{1}{T'+V_0} + \frac{(T'-t)+(T'+V_0)}{K+(T'-t)+(T'+V_0)} \right\} dK - \left\{ 1 + \frac{(T'-t)+(T'-t)+(T'+V_0)}{K+(T'-t)+(T'+V_0)} \right\} dK - \left\{ 1 + \frac{(T'-t)+(T'+V_0)}{K+(T'-t)+(T'+V_0)} \right\} dK - \left\{ 1 + \frac{(T'-t)+(T'+T'+V_0)}{K+(T'-t)+(T'+T'+V_0)} \right\} dK - \left\{ 1 + \frac{(T'-t)+(T'+T'+V_0)}{K+(T'+T'+V_0)} \right\}$$

(3.6)

Modern reversing thermometers have a minimum scale division of  $0.1^{\circ}$ C for its main thermometer. With the use of a magnifying glass, an experienced oceanographer is able to read the main thermometer to an accuracy of  $\pm 0.01^{\circ}$ C. The minimum scale division of an auxiliary thermometer is  $0.5^{\circ}$ C and it can be read to an accuracy of  $\pm 0.1^{\circ}$ C. The accuracy of V<sub>o</sub> value, which is a constant for an individual thermometer may be assumed to be  $\pm 0.5^{\circ}$ C. The accuracy of K may be taken as  $\pm 50^{\circ}$ C assuming only two significant figures in 6100. Using the above magnitudes of errors in the different independent quantities, the error due to the addition of expansion correction may be computed using the expression (3.6). Since K is assumed to be correct only to two significant figures, C and therefore  $\mathbf{d} \in \mathbf{C}$  will, in general, be correct only to two significant figures.

Table II(A) gives values of dC correct to 3 decimal places for different values of (I'-t) and (I'+V<sub>0</sub>). Fig. 1(A) shows the graph of dC against (I'-t) for two extreme values of (I'+V<sub>0</sub>), namely 70°C and 200°C. The graph shows that dC increases as both (I'-t) and (I'+V<sub>0</sub>) increases. The following generalisations can be made from this graph.

1) When  $(T'-t) < -4^{\circ}C$ , the value of  $dC < 0.005^{\circ}C$  for all values of  $(T'+V_{o})$ . dC may be neglected in such situations, and the maximum error in the corrected reading of the main thermometer,  $dT_{w}$ , is

$$d'T_{w} = dT + dC$$
$$= 0.01^{\circ}C + C$$
$$= 0.01^{\circ}C$$

	260°U	   4	ß	5	ω	0	IC	11	12	13	14	15
Let of $\mathbf{\dot{A}}$ C (i. 10 <sup>-3</sup> degrees C) for different values of (7'-t) and (7'+V <sub>0</sub> )		1   2]	£	Ś	7	ထ	6	10	11	12	14	ו   ד 
	13000	1 1 7	Ŋ	Ś	7	Ω,	0	10	11	12	13	14 14
	170°3	; [ 7]	വ	9	7	7	ω	6	10	11	25	E E
	150°C	1	4	ιΩ	ŝ	7	co	6	OT	11	12	
	'+V。) 150°C	1 1 1 1 1 1 1	マ	പ	Ś	7	ω	ሪ	0	OT	11	1.2
	1/00	: [ က]	<b>5</b> ¦-	Ŋ	\$	9	7	ω	တ	01	ЛC	י 1] 1
	130°C	၊ က ၊	4	4	۲ <u>۲</u>	Ś	-1	7	Ø	Ċ	ы	11
	120°C	1 1 1	ო	4	ເດ	5	S	7	co	co	σ	10
		i t ()	റ	4	4	ß	١D	Ĺ	7	ŝ	() )	6 1
	100 <sup>0</sup>	ຸ ຕ	ო	4	4	ĿΩ	ŋ	¢	7	7	ာ	۲ م ۱
	°006	1 1 (N) 1	ო	с	4	ケ	ഹ	5	9	7	7	I C
		[ 	2	ო	ო	4	:D	ß	9	5	7	۰ ۲
Valu	) 70 <sup>0</sup> ,	: [	2	ო	ო	4	Š.	ເດ	ιΩ	<u>`0</u>	\$	
		! ○ 	2°4	೦ <u>,</u> ೭	1000	1300	16°C	1 <sup>,0</sup> 0	22°C	25°C	28°0	3100





Fig. 1 (A): Curves of errors due to addition of expansion correction in a protected reversing thermometer for two extreme values of  $(T^{\bullet}+V_{o})$ 

Fig. 1 (E): Error due to addition of exca nsion correction for an individual protected reversing thermometer with  $V_0 = 105^{\circ}C$  at  $t = 27^{\circ}C$ . ii) When  $(T'-t) > -22^{\circ}C$ , the value of  $dC > 0.005^{\circ}C$  for all values of  $(T'+V_{o})$ . In this case, the maximum error in the corrected reading of the main thermometer,  $dT_{w}$ , should be obtained as

$$d^{r}T_{w} = dT' + dC$$
  
= 0.01°C+0.01°C  
= 0.02°C

iii) When (T'-t) is in the range from  $-4^{\circ}$ C to  $-22^{\circ}$ C,  $d^{\circ}T_{W}$  may be either 0.01°C or 0.02°C depending on the value of  $(T'+V_{c})$ .

As an example of finding the graph of  $d^{\circ}C$  against (f'-t) (and f') for an individual thermometer, for a specified value of t, let us assume that the  $V_0$  value of the thermometer is  $105^{\circ}C$  and t =  $27^{\circ}C$ . Now using fable II(A) and interpolating for the in-between values of (f'+ $V_0$ ), we get the required data for the above graph which is given in Table T(3). The corresponding graph is shown as Fig. 1(3). We get more or less the same graph for  $V_0$  values in the range from  $95^{\circ}C$  to  $115^{\circ}C$ , the common range of  $V_0$  for modern thermometers, and for t in the range from  $25^{\circ}C$  to  $29^{\circ}C$ , the common range of temperature in the tropical seas. The graph shows that for values of (f'-t) below about  $-13^{\circ}C$  (and for values of f' above

# Table II(B)

Values of dC (in  $10^{-3}$  degrees C) for different values of (T'-t), when V<sub>0</sub> =  $105^{\circ}$ C and t =  $27^{\circ}$ C

	 1°C	 4 <sup>0</sup> С	7°C	10°C		16 <sup>0</sup> C	19 <sup>0</sup> C	22°C	25°C	28 <sup>0</sup> C
d <sub>c</sub>	3	4	4	5	5	6	7	7	8	8

about  $14\pm2^{\circ}$ C), the error introduced due to the addition of expansion correction  $d^{\circ}$ C < 0.005°C, and so is negligible. For values of (T'-t) above  $-13^{\circ}$ C (and for values of T' below  $14\pm2^{\circ}$ C),  $d^{\circ}$ C > 0.005°C and so should be taken as 0.01°C. In a tropical ocean (f'-t) equal to  $-13^{\circ}$ C (and f' equal to  $14\pm2^{\circ}$ C) is usually found between the depths of 150 m and 200 m. In the treatment that follows, we will assume a tropical ocean where t can be taken as  $27\pm2^{\circ}$ C, and a thermometer having V<sub>0</sub> equal to  $105\pm10^{\circ}$ C. In this case, the error introduced due to the addition of expansion correction may be taken as negligible upto 200 m and as  $0.01^{\circ}$ C below 200 m. This means that the maximum error in the corrected thermometer reading is  $\pm0.01^{\circ}$ C upto 200 m and  $\pm0.02^{\circ}$ C below 200 m.

### 3.2. Pressure

The practical unit of measurement for pressure used in Physical Oceanography in 1 decibar which is equal to  $10^5$  dynes/cm<sup>2</sup> (Sverdrup et al. 1942). The pressure exerted per square centimetre by 1 metre of sea water is very nearly equal to 1 decibar and so the depth in metres and the pressure in decibars are expressed by nearly the same numerical values. Generally the former is slightly greater than 1 decibar but the difference dP', given by

d P' =  $\rho_g \propto 100 \text{ dynes/cm}^2 - 1 \text{ decibar}$ 

is usually less than 0.6% in a tropical sea. dP' is smaller and may even become negative in the surface layers of the ocean for latitudes towards the equator because both acceleration due to gravity and density are smaller under these conditions. Larger differences occur in deeper layers at higher latitudes because, both acceleration due to gravity and density become higher at these depths and latitudes.

A rough estimate of the depth is obtained by measuring the length of the wire rope payed out and the angle made by the wire with the vertical. This will give the true value of depth only when the depth and the wire angle are small. When both are large, the usual method of determining the depth is by comparing the corrected reading of the protected reversing thermometer with the corrected reading of an unprotected reversing thermometer. For the unprotected thermometer also we have to algebraically add the expansion correction and the index correction to get the corrected reading. The expression for the expansion correction for the unprotected reversing thermometer is given by

$$C_{u} = \frac{(f_{w} - t_{u}) (T'_{u} + Vo_{u})}{K_{u}}$$
(3.7)

where the symbols have the following meanings:

C<sub>u</sub> - expansion correction to be added algebraically to the main thermometer reading of the unprotected reversing thermometer, in <sup>o</sup>C

$$T_w \sim corrected reading of the protected reversing thermometer, in oC$$

- I'u main thermometer reading of the unprotected
   reversing thermometer, in <sup>O</sup>C
- Vou- volume of mercury below O<sup>O</sup>C mark of the main thermometer of the unprotected reversing thermometer, expressed in <sup>O</sup>C of the main thermometer scale
- $K_u$  reciprocal coefficient of the relative expansion of mercury and thermometer glass. This value for most of the modern thermometers is  $6100^{\circ}$ C.

The index correction,  $I_u$ , is for the errors caused by the irregular cross section and for the irregularities of the scale etching. Hence the total correction

$$\Delta T_{u} = C_{u} + I_{u} \tag{3.3}$$

and the corrected reading of the unprotected thermometer is

$$T_{u} = T'_{u} + \Delta T_{u}$$
$$= T'_{u} + C_{u} + I_{u}$$
(3.9)

Since all the different quantities in the expression (3.8) are subject to measurement errors, addition of the expansion correction and index correction to the uncorrected main thermometer reading introduces errors into the corrected main thermometer reading and is given by

$$d I_u = d I'_u + d C_u + d I_u$$
 (3.10)

The error  $dI_u$  is neglected for the same reasons as was discussed in the case of the protected reversing thermometer.  $dC_u$  is obtained by taking the total differential of the expression for  $C_u$  and is given by

$$d^{\prime}C_{u} = \begin{bmatrix} d^{\prime}T_{w} \\ T_{w} = -t_{u} \\ + \frac{d^{\prime}V_{u}}{T_{u} + V_{u}} \\ + \frac{d^{\prime}K_{u}}{T_{u} + V_{u}} \\ - \frac{d^{\prime}K_{u}}{R_{u}} \\ - \frac{d^{\prime}K_{u}}{R_{u}} \\ \end{bmatrix} C_{u}$$
(3.11)

Since each of the errors,  $dT_w$ ,  $dt_u$ ,  $dT_u^{\prime}$ ,  $dVo_u$  and  $dK_u$  are just as likely to be positive as negative, we should

take all the terms with a positive sign to be sure of the maximum error in  $C_u$ . Hence the final expression for  $dC_u$  is

$$\tilde{d} C_{u} = \begin{bmatrix} dT_{u} + dt_{u} \\ T_{w} + t_{u} \\ \end{bmatrix} + \begin{bmatrix} dT_{u} + dt_{u} \\ T_{u} + t_{u} \\ \end{bmatrix} + \begin{bmatrix} dT_{u} + dt_{u} \\ T_{u} + t_{u} \\ \end{bmatrix} + \begin{bmatrix} dT_{u} + dt_{u} \\ T_{u} \\ \end{bmatrix} + \begin{bmatrix} dT_{u} \\ T_{u}$$

The errors in  $f'_{u}$ ,  $t_{u}$ ,  $Vo_{u}$  and  $K_{u}$  are the same as the corresponding quantities in a protected thermometer. The errors in  $T_{w}$  have already been discussed in the preceding section. With these values for the different errors,  $dC_{u}$ may be computed using the expression (3.12).  $dC_{u}$  may be taken as correct to only two significant figures for the same reason as discussed in the preceding section.

Computation of  $\mathbf{d} C_u$  for different values of  $(T_w - t_u)$ and  $(f_u + Vo_u)$ , for both the cases when  $\mathbf{d} T_w$  is  $\pm 0.01^\circ$ C and  $\pm 0.02^\circ$ C, results in the same numerical values as shown in Table II(A), the maximum difference being a unit in the third decimal place, for higher values of  $(f_w - t_u)$  and  $(f_u + Vo_u)$ . Hence the graph for  $\mathbf{d} C_u$  against  $(f_w - t_u)$ for the two extreme values of  $(f_u + Vo_u)$ , namely  $70^\circ$ C and  $200^\circ$ C, will also be of the same shape and range as Fig. I(A), and so the generalisations arrived at from Fig. I(A) will hold good for an unprotected reversing thermometer also. Calculation of  $\mathbf{d} C_u$  for different values of  $(T_w - t_u)$  (and  $T_u'$ ) for an individual thermometer

\_44\_

for a specified value of t is rather cumbersome, since T' increases roughly by C.Ol<sup>O</sup>C for every 1 metre increase in depth. But it can easily be seen that since the difference between the protected and the suprotected thermometer readings is not large in the surface layers, a curve of  $dC_u$  drawn against  $(T_w - t_u)$  (and  $T'_u$ ) for a V value in the range from  $95^{\circ}C$  to  $115^{\circ}C$  and for t<sub>1</sub> in the range from 25°C to 29°C will have the same shape and range as in Fig. I(B). For higher values of  $(T_w-t_u)$  (and for lower values of  $T_{ij}^{t}$ , the difference in the readings of the unprotected and the protected thermometers increases and consequently the curve for  $\hat{\mathcal{J}}^{\mathsf{C}}\mathsf{C}_{\mathrm{u}}$  shows, slightly higher values than the values shown by Fig. I(B). Hence we may reasonably assume that the error in the corrected reading of the unprotected thermometer is <u>+</u>0.01°C upto a depth of 200 metres and +0.02°C below 200 metres.

Depth is calculated using the formula (LaFond, 1951)

$$D = \frac{T_u - T_w}{Q / m}$$
(3.13)

where the symbols have the following meanings D - depth in metres  $T_u - corrected$  unprotected thermometer reading in  $^{O}C$ 

$$T_w$$
 - corrected protected thermometer reading in  $^{O}C$   
 $p_m$  - mean density of the water column above the depth of reversal of the thermometers

Q - pressure coefficient of the unprotected thermometer expressed in <sup>O</sup>C increase in the realing per 0.1 kg/cm<sup>2</sup> increase in pressure.

The error in the result using the above formula due to the errors in the different quantities may be obtained by taking the total differential of the expression.

$$d D = \begin{bmatrix} dT_{u} \\ (T_{u} = T_{w}) - (T_{u} = T_{w}) - \frac{dQ}{(T_{u} = T_{w})} - \frac{dQ}{Q} - \frac{dP_{m}}{P_{m}} \end{bmatrix} D$$
(3.14)

Since each of the errors  $d_{T_u}$ ,  $d_{T_w}$ ,  $d_{Q}$  and  $J'_m$  are just as likely to be positive as negative we should take all the terms with a positive sign in order to be sure of the maximum error in D. Hence

$$d D = \begin{bmatrix} dT_u + dT_w \\ T_u - T_w \end{bmatrix} + \frac{dQ}{Q} + \frac{dP_m}{P_m} \end{bmatrix} D$$
(3.15)

We have already discussed  $d^{T}T_{u}$  and  $d^{T}T_{w}$  and found that their magnitude is +0.01°C upto a depth of 200 metres and ±0.02°C below 200 metres. Q and  $P_{m}$  are usually known correct to 4 significant figures and 6 significant figures respectively and so, the magnitudes of  $\frac{d^{2}}{Q}$  and  $\frac{d^{2}m}{p^{m}}$ will be small compared to the term,

$$\frac{d}{T_u} \frac{T_u + d}{T_u - T_w}$$

Hence the former two terms may be neglected. The expression for **d** D may therefore be approximated as

$$\dot{d} D = \begin{bmatrix} \dot{d} \tau_u + \dot{d} T_w \\ T_u - T_w \end{bmatrix} D$$
(3.16)

The quantity  $Q \int_{m}^{C}$  is approximately equal to  $0.01^{\circ}$  C/metre (LaFond, 1951) and D may be written as

$$D = \frac{T_{u} - I_{w}}{Q_{m}} = \frac{T_{u} - T_{w}}{0.01^{\circ} C/m}$$
(3.17)

Hence, 
$$dD$$
 is given by  
 $d D = \begin{bmatrix} dT_u + dT_w \\ (T_u^- - T_w^-) \end{bmatrix} D$   
 $= \frac{dT_u + dT_w}{(T_u^- - T_w^-)} \times \frac{(T_u - T_w)}{0.01^\circ C/m}$   
 $= \frac{dT_u + dT_w}{0.01^\circ C/m}$ 

=  $|(d f_u + dT_w)| 100|$  metres (3.18)

The sum  $(d^{\dagger}T_{u} + dT_{w})$  will be  $0.02^{\circ}$ C upto 200 metres depth and  $0.04^{\circ}$ C below 200 metres depth. So the error in  $d^{\dagger}D$ may be taken as 2 metres in absolute magnitude upto a depth of 200 metres and 4 metres in absolute magnitude below 200 metres. Hence the total error in taking the pressure in decibars as numerically equal to depth in metres is the sum of dP' and the pressure equivalent of the error in depth dD. Thus, for example, the maximum error in taking the pressure at 1000 metres depth equal to 1000 decibars is about 10 decibars in the tropical seas. The actual value is usually lesser since, as mentioned earlier, dP' in the surface layers of the ocean, particularly at latitudes near the equator, is small.

#### 3.3. Salinity

Water dissolves almost all known substances and in greater quantities compared to any other liquid. Determination of the total quantity of the dissolved substances is necessary since this affects the density of a water sample. Because of the complexity of sea water, direct determination of the total dissolved substances by chemical means is a near impossibility. Also it is not possible to obtain reproduceable results by evaporating sea water to dryness and weighing the residue since some of the materials present, chiefly chlorides, are lost in the last stages of drying. These difficulties were overcome by following a technique yielding reproduceable results which, although do not represent the total quantity of dissclved solids, do represent a quantity, closely related and called the salinity of water. This

-48-

technique was established by an International Commission (Knudsen, 1901; Knudsen et al. 1902; Forch et al. 1902) and on the basis of its work salinity was defined as 'the total amount of solid material in grammes contained in one kilogramme of sea water when all carbonates have been converted to oxides, the bromine and iodine replaced by chlorine and all organic matter completely oxidised'. The salinity obtained according to this definition is slightly less than the total salt content. In fact, a water having a salinity of 34.32%. will have a total salt content of 34.48%. (Lyman and Fleming, 1940). Determination of salinity by the method of the International Commission is very rarely carried out because it is time consuming and difficult (Forch et al. 1902; Sorensen, 1902). Also it is subject to the following two sources of errors (Riley, 1975).

 a) Hydrogen chloride is lost through hydrolysis of magnisium chloride

b) water of crystallisation is retained tenaciously

Morris and Riley (1964) describes a method for the gravimetric determination of salinity which eliminates the above sources of errors. This method, eventhough yields very accurate results with careful work, is not suitable

-49-

for routine determinations as it is also time consuming.

Since Dittmar's analysis of 77 water samples collected during the Challenger Expeditio (Dittmar, 1884), it became obvious that regardless of the absolute concentrations, the relative proportions of the different major constituents are virtually constant. This concept of constancy of composition of sea water is only slightly changed after more precise analyses of numerous sea water samples for the major constituents (Cox et al. 1967). Murray (1893) suggested that it is only necessary to determine the chlorine content in a definite weight of water to ascertain the respective quantities of other constituents. The total halogen content in a definite quantity of water is expressed in terms of chlorinity which is defined as the total amount of chlorine, bromine and iodine in grammes contained in one kilogramme of sea water assuming that the bromine and iodine have been replaced by chlorine (Forch et al. 1902; Sorensen, 1902). A relationship between salinity and chlorinity, namely,

$$S_{*} = 0.030 + 1.8050 \text{ cl}$$
 (3.19)

was established by Knudsen (Knudsen et al. 1902) from the the salinity data of Sorensen (1902) and this equation

-50-

became the working definition of salinity. One drawback of the above definition is that it changes every time the atomic weights were changed. Because of this problem, chlorinity was defined in the following way (Jacobsen and Knudsen, 1940): 'The number giving the chlorinity, in per mille, of a sea water sample is, by definition, identical with the number giving the mass with unit gramme of atomic weight silver just necessary to precipitate the halogens in 0.3285234 kilogramme of sea water sample'. This definition of chlorinity, being independent of any redetermination of atomic weights, has been used by all oceanographers and the chlorinity titration remained the preferred precision method for its determination until very recently.

Since mid-fifties, this working definition of salinity come to be questioned increasingly for three reasons.

1. Development of conductimetric salinometers established that the chlorinity determination was no longer the only standard method for determining the salinity of sea water.

2. New investigations of the gravimetric salinity determination (Morris and Riley, 1964) threw doubt on the absolute accuracy of Sorensen's methods.

3. The small number and the non-representative distribution of sea water samples used by Sorensen cast doubt on the validity of Knudsen's relationship between salinity and chlorinity.

Because of these factors, a new investigation of the interrelationship between the measured parameters (chlorinity, conductivity ratio) and the derived parameters (salinity, sigma-t) was conducted and an International Joint Panel made the following recommendations (UNESCO, 1962,1965,1966,1968; Cox et al. 1967):

1. Salinity be defined in terms of conductivity ratio at  $15^{\circ}$ C relative to the conductivity of 'standard sea water' having a salinity of 35%. and chlorinity of 19.375%. This means that the position of importance held by the chlorinity titration for the determination of salinity was removed and conductivity measurement took its place. Since the 'standard sea water' is prepared assuming constancy of composition of sea water, which is only nearly true (Cox <u>et al.</u> 1967), the new definition of salinity also is dependent on this concept (Culkin, 1965). This concept can be avoided only when salinity is defined in terms of absolute conductivity instead of relative conductivity.

-52-

2. The relationship between salinity and chlorinity be redefined by a truely proportional relationship, namely,

5%. = 1.80655 Cl%. (Mooster <u>st</u> <u>al.</u> 1969) (3.20) When chlorinity is determined by titration, it be reported as chlorinity itself and use the above relationship to get an estimate of salinity.

The above relationship is compatible with the Knudsen relation and these two yield identical results at a salinity of 35%.. At 32%. and 38%. salinities, the difference is 0.0026%. and at 6%. salinity it is 0.025%. (Lyman, 1969). This new definition of chlorinity is an arbitrary one and does not have a chemical meaning, but the salinity obtained this way is very close to the total salt content of sea water (Grasshoff, 1976).

Thus the two methods that are used for the routine determination of salinity of sea water are the following:

- a) Chlorinity titration
- b) Conductivity ratio measurement

#### a) Chlorinity titration

Salinity letermination by means of chlorinity titration, even nough old fashioned, is not completely outdated. One region for this is that many small laboratories do not have the relatively expensive equipment needed for the conductometric determination of salinity. Another reason is that only small amount of sample is needed for the chlorinity titration compared to the conductivity measurement (Grasshoff, 1976).

The chlorinity of sea water is determined using Mohr-Knudsen method of titration in which the halogen ion in the sea water is titrated with silver nitrate using pottasium chromate as indicator. The silver nitrate solution used should be standardised against 'standard sea water'.

There are several sources of errors in this method.

1. The error caused in pipetting the sea water sample is  $\pm 0.01\%$ . in chlorinity on a chlorinity of 19%. (Riley, 1965).

2. A temperature change by  $2^{\circ}$ C of the silver nitrate solution causes an error of  $\pm 0.01\%$ . in chlorinity (Riley, 1965).

3. The smallest division of a Knudsen burette is 0.01 milli litre. The reading accuracy of such a burette is only 0.005 milli litre, equivalent to 0.005%. in chlorinity, since the meniscus will interfere with the reading (Fomin, 1964).

4. A similar error of 0.005%. in chlorinity **is** caused in the standardisation titration of the silver nitrate solution.

With correct calibration of the burette, proper care and experience, a standard deviation of 0.01/. in chlorinity can be obtained (Grasshoff, 1976). Hence the error in the determination of chlorinity by the above method can be taken as  $\pm 0.02/.$  at 95% confidence level. The corresponding error in salinity is obtained by taking the total differential of the equation (3.20),

as 
$$d_{5/.} = 1.80655 d_{C1/.}$$
  
=  $\pm 0.036/.$   
 $- 0.04/.$ 

### b) Conductivity ratio measurement

As already mentioned, the salinity determination by conductivity ratio measurement has almost completely superseded the chemical determination of chlorinity because of the increased speed, reliability and simplicity of modern salinometers and because of the fact that they can be used at sea by relatively unskilled personnel. Although Knudsen (Knudsen <u>et al</u>. 1902; Knudsen, 1903) had tried measurement of conductivity as a means to determine salinity more than 80 years ago, only recently, with the development of reliable electronic equipments (Brown and Hamon, 1961), could the method be effectively used for this purpose.

By international agreement (Wooster <u>et al.</u> 1969), the conductivity ratio measured by a salinometer is converted to a salinity value by the use of Tables (U.NESCO, 1966) which have been compiled from calculation of salinities from conductivity ratios at  $15^{\circ}$ C using the fifth order polynomial obtained from the data by Cox <u>et al.</u> (1967) after adjusting the constant term by the addition of a small quantity (+0.00018) to make  $R_{15} = 1.00000$  exactly corresponding to a salinity of 35.000/...

$$S_{..} = -0.08996 + 28.29720 R_{15} + 12.80832 R_{15}^{2}$$
$$-10.67869 R_{15}^{3} + 5.98624 R_{15}^{4}$$
$$-1.32311 R_{15}^{5}$$
(3.22)

If the conductivity ratio is measured at a temperature other than  $15^{\circ}C$ , the conductivity ratio  $R_t$  is converted to

-56-

 $\rm R_{15}$  by the addition of a correction term  $\Delta_{15}$  (t) given by the equation

$$\Delta_{15}(t) = R_{15} - R_{t} = 10^{-5} R_{t}(R_{t}-1) (t-15)$$

$$[96.7-72.0 R_{t} + 37.3 R_{t}^{2}$$

$$-(0.63+0.21 R_{t}^{2} (t-15)] \qquad (3.23)$$

This correction term is also obtained from the UNESCO Tables.

Instrument manufactures of the bench type salinometers usually claim a reproduceability of 0.003%. to 0.002%. in salinity, if the measurement is carefully performed. But the accuracy of salinity determination by this method is much poorer under realistic field work conditions, even assuming no systematic errors, for the following reasons.

1) Investigation by Grasshoff and Hermann (1975) has shown that the comparability of salinity measurements performed with different instruments, even of the same type, is much worse than the reproduceability of an individual instrument, especially if the salinities are not close to the point of calibration. This source of error is partially overcome by introducing 'standard sea water' of different salinities.

2) Samples used for the measurements by Cox et al. (1967), which form the basis for the International Oceanographic Tables, Vol.I, and the 'standard sea water' have been standardised with respect to the carbonate system. In fact the composition of sea water is liable to changes in the carbonate system. Park (1965) and Grasshoff (1968) have demonstrated the effect of pH (as a measure of the state of the carbonate system) on the relative conductivity and have shown that the salinity, obtained by measurement of conductivity ratio, changes as pH changes, without changing the total salt content of the seawater. They have also demonstrated that this apparent change in salinity is largest in the pH range occurring in the natural sea water and for high salinities. The above studies have also shown the effect of other variable micro constituents on the apparent salinity of sea water, but their influence has less dominance compared to the changes in the carbonate system.

Considering the above sources of errors, it will be unrealistic to assume in the salinity determination by the measurement of conductivity ratio, an accuracy better than 0.02/.. under realistic field work conditions (Grasshoff, 1976). The corresponding error in the chlorinity value may be taken as  $\pm 0.01/..$ 

### 3.4. On the accuracy of in situ salinometers

Accurate conductivity measurements in situ present difficulties additional to those experienced in the laboratory environment. Conductivity is a function, not only of salinity of sea wate: but also of temperature and pressure and these two parameters must be monitored for compensation purposes. The instrument which measures salinity (or conductivity), temperature and pressure in situ are known as STD and CTD probes. The former has an internal compensation circuit to produce a value for salinity directly from the three measured parameters.

There are several possible sources of errors in the data obtained using the SfD and CTD probes.

1. When the time constants of the sensors differ, the compensation circuit of the STD probe produces characteristic transient spikes if a strong temperature gradient is present (Goulet and Culverhouse, 1972). In this situation, it is better to employ a CfD probe and to apply compensation for time constant differences during the calculation of salinity.

2. Another cource of error in the STD probe data is the sudden jump in the salinity calibration which usually

-59-

occurs due either to a mechanical shok to the sensor head or to a dry solder joint in the probe electronics (Wilson, 1975). They are normally of the order of 0.1%. in salinity and may pass unnoticed unless a close watch is kept. Such calibration jumps can occur for the other parameters also. A usual remedy is to take at least one reversing bottle sample with each STD cast, the bottle being attached to the probe unit or just above it and triggered by messenger after equilibration of the reversing thermometers.

3. The conductivity obtained from the data of **CTD** probe is only a relative value, relative to the calibration of the instrument. Hence it is important to ascertain the conditions of salinity, temperature and pressure under which the unit was calibrated and also the value of absolute conductivity which was assumed for these conditions. Since the latter is not always explicitly stated, confusion on this point can lead to a significant error (walker and Chapman, 1973).

4. A possible source of error in the computation of salinity from the conductivity - temperature - depth data from a CTD probe is from the equation used for the purpose. For example, equation for the computation of salinity from conductivity ratio and temperature given by Cox et al.(1967) is true only in the range of temperature from  $10^{\circ}$ C to  $30^{\circ}$ C, although work is in progress to extend the range to  $0^{\circ}$ C (Milson, 1975). The equation given by Perkin and Walker (1972) for the same purpose is true only in the range of temperature from  $-2^{\circ}$ C to  $20^{\circ}$ C. These equations show a marked deterioration outside the stated ranges. The most widely used relationship for the pressure correction is that of Bradshaw and Schleicher (1965).

Hence it is evident that the present SID and CTU units, eventhough highly useful, are, by no means, ideal. Besides the problems of 'spiking' and calibration jump in the STD probes and the non-standardised calibration procedures and differences in calculation in the CDD probes, the available units are not very much reliable and tend to be too complicated and expensive. Some of the problems can be eliminated by taking a reversing water bottle sample during the cast which will allow a correction to be applied to the results from the in situ probe to bring them into agreement with the International Uceanographic Tables. Hence the accuracy of an in situ probe can never be better than that obtainable from a reversing water bottle sample. This situation will continue until instrumental stability is greatly improved, reliable absolute conductivity values are available and

-61-

the validity of the relationship to calculate salinity in the International Tables is extended below 10°C. Only in such a situation can the experical correction of the CFD and STD data using a reversing water bottle and reversing thermometers can be discarded.

#### CHAPTER IV

## ERRORS IN THE COMPUTATION OF DEPENDENT OCEANOGRAPHIC VARIABLES

In the preceding chapter we have discussed the random errors that occur in the measurements of the independent oceanographic variables, namely, temperature, depth and salinity. The values of these variables are used to calculate the dependent oceanographic variables, namely, density <u>in situ</u>, sigma-t, specific volume and specific volume anomaly. Since the measurements of the independent variables are subject to random errors, the resulting dependent variables will also be in error, the magnitude of which may be obtained using the law of propagation of errors.

# 4.1. Density in situ and sigma-t

The density of any substance is defined as the mass per unit volume and in the cgs system it is stated in grammes per cubic centimetre. The specific gravity of any substance is defined as the ratio of its density to that of distilled water at  $4^{\circ}$ C and under atmospheric pressure. Since in the cgs system the density of distilled water at  $4^{\circ}$ C is unity, the specific gravities are numerically
identical with densities. In oceanography, the term density is generally used eventhough specific gravity is the one that is meant.

The density of sea water depends upon the three independent variables, namely, temperature, salinity and pressure. The density of a sea water sample at the temperature and pressure at which it was collected, known as the density <u>in situ</u>, is denoted by the symbol  $\rho_{s,t,p}$ . Knudsen (1901) introduced, for the purpose of convenient handling of numerical values, a symbol  $\sigma_{s,t,p}$  as a measure of density <u>in situ</u> defined as

$$\sigma_{s,t,p} = (\rho_{s,t,p}-1) \ 10^3 \tag{4.1}$$

The corresponding quantity at atmospheric pressure is written as  $\sigma_t$  and that at atmospheric pressure and  $O^OC$  as  $\sigma_0$ .  $\sigma_t$  may be obtained from the relation (Knudsen, 1901)

$$\sigma_{t} = -\Sigma_{t} + (\sigma_{0} + 0.1324) [1 - A_{t} + \beta_{t} (\sigma_{0} - 0.1324)] (4.2)$$

where

$$\Sigma_{t} = \frac{(T-3.98)^{2}}{503.570^{-1}} \times \frac{(T+283)}{(T+67.26)}, \qquad (4.3)$$

$$\sigma_{0} = -0.039 + 1.470 \, \text{s} \, \text{Cl-0.001570 Cl}^{2} + 0.0000398 \, \text{Cl}^{3}, \quad (4.4)$$

$$A_{t} = [4.7867 \ f - 0.098185 \ T^{2} + 0.0010843 \ f^{3}] \ 10^{-3},$$
(4.5)

and

$$B_{t} = [18.030T - 0.8164 T^{2} + 0.01667 T^{3}] 10^{-6}$$
(4.6)

 $\sigma_{s,t,p}$  as a measure of density <u>in situ</u> may be obtained by the addition of the correction terms computed from formula established by Ekman (1908).

The error in the result of computation for  $\sigma_t$  due to the errors in the temperature and chlorinity measurements may be obtained by taking the total differential of the expression given above and the result may be written as follows.

$$d\sigma_{t} = \alpha \ dCl + \beta dT \tag{4.7}$$

where

$$\alpha = [1.4708 - 0.003140 C1 + 0.0001194 C1] \times [1-A_t + 2 \sigma_0 B_t]$$
(4.3)

$$\beta = (\sigma_0 + 0.1324) [(\sigma_0 - 0.1324) (18.030 - 1.6328 T + 0.05001 T^2) 10^{-6} - (4.7867 - 0.196370 T + 0.0032529 i^2) 10^{-3}] - (4.7867 - 0.196370 T + 0.0032529 i^2) 10^{-3}] - (4.7867 - 0.196370 T + 0.0032529 i^2) 10^{-3}]$$

$$\Sigma_{t} \left( \frac{2}{1-3.98} + \frac{1}{1+283} - \frac{1}{1+67.26} \right)^{*}$$
(4.9)

The coefficients  $\alpha$  and  $\beta$  calculated for different values of temperature and chlorinity are shown as Table III(A) and Table III(B) respectively. Table III(B) shows that  $\beta$  Can be either positive or negative. But, since the error dT is just as likely to be positive as negative, we should take the term  $\beta$  dT with a positive sign to be sure of the maximum error in  $\sigma_t$ .

1

In chapter 3 we have discussed the magnitudes of the random errors in the measurements of temperature and \*Fomin (1964) gives the expression for  $\beta$  as  $\beta = (\sigma_0 + 0.1324) \left[ (\sigma_0 - 0.1324) (18.030 - 1.6328 T + 0.05001 T^2) 10^{-6} - (4.7867 - 0.196370 T + 0.0032529 T^2) 10^{-3} - \Sigma_t (T = 3.98 + T = 1.283 - T = 1.283) \right]$ (4.10)

Here the position of the double bracket is at the end of the expression which is wrong. This is not a printing mistake because Fomin actually used the wrong expression for the computation of values of  $\beta$ .

chlorinity. It was established that the error in the measurement of temperature in a tropical sea should be considered as  $\pm 0.01^{\circ}$ C up to a depth of 200 m and as  $\pm 0.02^{\circ}$ C below 200 m. The temperature observed at a depth of 200 m in a tropical sea is around 13°C and so we can estimate the error committed in the measurement of temperature above  $13^{\circ}$ C to be  $\pm 0.01^{\circ}$ C and that below  $13^{\circ}$ C to be  $\pm 0.02^{\circ}$ C. We have also established that the error in the chlorinity determinations by the method of chlorinity titration is +0.02%. and that by the method of conductivity ratio measurement is ±0.01%.. Assuming as above the errors in the determination of temperature and chlorinity, the errors in the computed values of  $\sigma_{+}$  for different values of temperature and chlorinity may be obtained using expression (4.7) and Tables III(A) and III(B). It can easily be observed that for almost all values of temperature and chlorinity, the error in  $\sigma_+$  is greater than 0.015 when chlorinity is determined by the method of conductivity ratio measurement and that the error is around +0.03 when the chlorinity is estimated by the method of chlorinity titration. Hence in the discussion that follows we will assume that the error in the computation of  $\sigma_{\pm}$  is  $\pm 0.02$  when the chlorinity is determined by the method of conductivity ratio measurement and is +0.03

-67-

Table III(A)

Variation of coefficient  $\alpha$  with temperature and chlorinity

			19 Con Con (14 K2 62 64 64				) T		1			
	7.•-2		4	7	10	13	16	19	22	25	23	31
Ţ	1.45	1•45	<b>1</b> • 43	Ĺ•⊹]	<b>1.4</b> 0	1 <b>.</b> 39	1.38	I.37	1.36	1.35	1.34	1.34
2	1.46	1.44	1.43	1.41	1.40	1 <b>.</b> 39	1 <b>.</b> 38	1.37	1.36	1•35	1.34	1.34
14	<b>1.</b> 46	1.44	1.43	1.41	<b>1.</b> 40	1.39	<b>1.</b> 38	1•31	1.36	1.35	<b>1</b> .35	<b>1.</b> 34
16	<b>1.</b> 46	1.45	1.43	1.42	1.40	1.39	1.38	1°37	<b>L-</b> 36	1.36	1 <b>.</b> 35	<b>1.</b> 34
13	1.46	1.45	1.43	1.42	1.40	1.39	1.38	<b>1.</b> 37	1.37	1 <b>.</b> 36	<b>1.</b> 35	<b>1.3</b> 5
20	1.47	1.45	1.44	<b>1.</b> 42	1.41	<b>1.4</b> 0	1.39	<b>1.</b> 38	<b>1.</b> 37	1.36	<b>1.</b> 36	<b>1.</b> 35
22	1.47	1.45	1.44	].43	1.41	1.40	1 <b>.</b> 39	<b>1.3</b> 8	<b>1.</b> 38	1.37	1.36	1 <b>.</b> 36

I

1111

Table III(3)

Variation of coefficient  $\beta$  with temperature and chlorinity

ξ	7						<b>-</b>					
5	/2	1		7	10	13	16	19	22	25	28	31
10	+0.03	-0.01	-0.0-	-0.10	-0.13	-0.17	<b>-</b> 0 <b>.</b> 20	-0.23	<b>-</b> 0.26	<b>-</b> 0.28	-0.31	<b>-</b> 0•33
12	+0.02	-0.03	••0*0.	-0.11	<b>-</b> 0 <b>.</b> 14	-0.17	-0.20	<b>-</b> 0 <b>.</b> 23	<b>-</b> 0 <b>.</b> 26	-0.29	-0.31	<b>-</b> 0.34
14	+0.01	-0,04	<del>ن</del> اني، ن	~.0 <b>.1</b> 2	-0.15	-0.18	-0.21	-0.24	-0,27	-0.29	-0.32	-0.34
16	0.00	<b></b>	ő∵* <b>0-</b>	-0° 15	-C.16	-C, 19	-0.22	-0-25	-0.27	-0-30	<b>-</b> 0.32	<b>-</b> 0.34
13	-0.02	-0.06 -06	-0, 11	→( <b>.</b> 1£		-0,20	-0.22	-0.25	<b>-</b> 0 <b>.</b> 28	-0.30	<b>-0.</b> 32	<b>-</b> 0 <b>.</b> 35
20	-0.03	-0.07	<b>1</b> ° O <b>1</b>	- 0.14	5 T - O-	-0,20	-0.23	-0,26	<b>-</b> 0 <b>.</b> 28	-0.31	-0.33	<b>-</b> U.35
22	-0.04	<b>.</b> 08	0,12	-0,15	8°°°0+	0.21	-0.24	<b>-</b> 0 <b>.</b> 26	<b>-</b> 0.29	-0.31	<b>-</b> 0.33	<b>-</b> 0.35

5

na bha an an ann an tar an tar

when the chlorinity is determined by the method of chlorinity titration.

Cox et al. (1968) have pointed out that the absolute density of distilled water at 4°C and at atmospheric pressure is about 3 parts in 10<sup>5</sup> below unity and hence the absolute density of sea water is not quite equal to its specific gravity. They also pointed out that the distilled water used by Forch et al. (1902) in their specific gravity determinations was of unknown isotopic composition. Since the distillation procedure used by them is not given, it is now not possible to reproduce the standard used by Forch et al. (1902). This situation warranted a new investigation into the relationship between salinity, temperature and  $\sigma_+$ . Cox et al. (1970) have described an apparatus designed for this purpose. They found that Knudsen's tables gave slightly lower values of specific gravity over the salinity range of normal sea water and the deviation became larger for lower salinities. Investigations of Kremling (1971,1972) using a more accurate density comparison instrument, developed by Kratky et al. (1969), confirmed the above conclusions of Cox et al. (1970). Kremling's work showed that Knudsen's values were low, on an average, by 0.013 in  $\sigma_+$  in the salinity range from

9%. to 39%. and the deviation was about 0.025 in  $\sigma_{\rm t}$  at a salinity of 5%..

Using the data obtained from his experiments Cox <u>et al.</u> (1970) have developed an emperical relationship between temperature, salinity and  $\sigma_t$ , valid in the salinity range from 9%. to 41%., as:

$$\sigma_{t} = 8.00969062 \times 10^{-2} + 5.88194023 \times 10^{-2} T + 7.97018644 \times 10^{-1} \text{s} - 8.11465413 \times 10^{-3} T^{2} - 3.25310441 \times 10^{-3} \text{ sT} + 1.31710842 \times 10^{-4} \text{ s}^{2} + 4.76600414 \times 10^{-5} \Gamma^{3} + 3.89187483 \times 10^{-5} \text{ s}\Gamma^{2} + 2.87971530 \times 10^{-6} \text{ s}^{2} \text{T} - 6.11831499 \times 10^{-8} \text{ s}^{3}$$
(4.11)

As before, the error in the result of computation for  $\sigma_t$  due to the errors in the temperature and salinity measurements may be obtained by taking the total differential of the above expression and the result may be written as:

$$d\sigma_t = a \, ds + b \, dt \tag{4.12}$$

where

$$a = 7.9702 \times 10^{-1} - 3.2531 \times 10^{-3} T +$$
  
2.6342 × 10<sup>-4</sup> S + 3.8919 × 10<sup>-5</sup> T<sup>2</sup> +  
5.7594 × 10<sup>-6</sup> ST - 1.8355 × 10<sup>-7</sup> S<sup>2</sup> (4.13)

$$b = 5.8319 \times 10^{-2} - 1.6629 \times 10^{-2} T -$$
  
3.2531 × 10<sup>-3</sup> S + 1.4298 × 10<sup>-4</sup> T<sup>2</sup> +  
7.7838 × 10<sup>-5</sup> ST + 2.8797 × 10<sup>-6</sup> S<sup>2</sup> (4.14)

The coefficients a and b are calculated for different values of temperature and salinity and are shown as Table IV(A) and Table IV(B) respectively. Table IV(B)shows that b can be either positive or negative. But, since the error dT is just as likely to be positive as negative, we should take the term bdT with a positive sign to be sure of the maximum error in  $\sigma_+$ .

Considering a tropical sea and following the same argument discussed in connection with the computational error in  $\sigma_t$  when Knudsen's relation is used for the purpose, we can estimate that the error committed in the measurement of temperature above  $13^{\circ}$ C is  $\pm 0.01^{\circ}$ C and that below  $13^{\circ}$ C is  $\pm 0.02^{\circ}$ C. As established in section 3.3, the error in the determination of salinity by the chlorinity titration method is  $\pm 0.04\%$ . and that by the method of conductivity ratio measurement is  $\pm 0.02\%$ . Quantifying as above the errors in the determination of temperature and salinity, the error in the computed value of  $\sigma_t$  for different values of temperature and salinity may be obtained using the relation (4.12) and fables IV(A) and IV(3). Here also it may be observed Table IV(A)

Variation of coefficient 'a' with temperature and salinity

T         T           2         1         4         7         1C         13         16         19         22         25         28         31           .81<0.80<0.79<0.78         0.77         0.77         0.76         0.75         0.75         0.74         0.74           .81<0.80<0.79<0.78         0.77         0.77         0.76         0.75         0.75         0.74         0.74           .81<0.80<0.79<0.78         0.77         0.77         0.76         0.76         0.75         0.74         0.74           .81<0.80<0.79<0.79         0.77         0.77         0.76         0.76         0.75         0.75         0.75           .81<0.80<0.79<0.79         0.78         0.77         0.77         0.77         0.76         0.75         0.75         0.75           .81<0.80<0.79<0.79         0.79         0.77         0.77         0.76         0.76         0.75         0.75         0.75           .81<0.80<0.79<0.79         0.79         0.77         0.77         0.76         0.76         0.75         0.75         0.75           .81<0.80<0.79         0.79         0.77         0.77         0.76         0.76         0.75         0.75         <				<u>6</u>					
2         1         4         7         1C         13         16         19         22         25         28         31           .81         0.80         0.79         0.78         0.77         0.77         0.76         0.75         0.75         0.74         0.74           .81         0.80         0.79         0.78         0.77         0.77         0.76         0.75         0.75         0.74         0.74           .81         0.80         0.79         0.78         0.77         0.76         0.75         0.75         0.74         0.74           .81         0.30         0.79         0.78         0.77         0.76         0.76         0.75         0.75         0.74           .81         0.80         0.79         0.78         0.77         0.71         0.76         0.76         0.75         0.75         0.75           .81         0.80         0.79         0.77         0.77         0.76         0.76         0.75         0.75         0.75         0.75         0.75           .81         0.80         0.79         0.77         0.77         0.76         0.76         0.75         0.75         0.75         0.75				-	•				
81       0.80       0.77       0.77       0.77       0.76       0.75       0.75       0.74       0.74         81       0.80       0.79       0.77       0.77       0.76       0.75       0.74       0.74       0.74         81       0.80       0.79       0.77       0.77       0.76       0.75       0.74       0.74       0.74         81       0.30       0.79       0.77       0.77       0.76       0.75       0.75       0.75       0.74       0.74         81       0.80       0.79       0.77       0.77       0.76       0.75       0.75       0.75       0.75       0.75         81       0.80       0.79       0.77       0.77       0.76       0.76       0.75       0.75       0.75         81       0.80       0.80       0.76       0.77       0.77       0.76       0.76       0.75       0.75       0.75         81       0.80       0.80       0.77       0.77       0.76       0.76       0.75       0.75       0.75       0.75       0.75       0.75       0.75       0.75       0.75       0.75       0.75       0.75       0.75       0.75       0.75 <t< th=""><th>2 1 4 7</th><th>lC</th><th>13</th><th>16</th><th>19</th><th>22</th><th>25</th><th>28</th><th>31</th></t<>	2 1 4 7	lC	13	16	19	22	25	28	31
81         0.30         0.77         0.77         0.76         0.75         0.75         0.74         0.74         0.74           .81         0.30         0.79         0.78         0.77         0.77         0.76         0.75         0.75         0.75         0.74         0.74         0.74           .81         0.30         0.79         0.78         0.77         0.77         0.76         0.75         0.75         0.75         0.74           .81         0.80         0.79         0.78         0.77         0.77         0.76         0.75         0.75         0.75         0.75           .81         0.80         0.79         0.78         0.77         0.77         0.76         0.76         0.75         0.75         0.75           .81         0.80         0.79         0.78         0.77         0.77         0.76         0.76         0.75         0.75         0.75         0.75	.81 0.80 0.79 0.78	0,77	0.77	0.76	0.75	0.75	0.75	0.74	0.74
.81       0.30       0.78       0.76       0.76       0.75       0.75       0.75       0.75         .81       0.80       0.79       0.76       0.77       0.77       0.77       0.76       0.75       0.75       0.75         .81       0.80       0.79       0.76       0.77       0.77       0.77       0.76       0.75       0.75       0.75         .81       0.80       0.79       0.77       0.77       0.77       0.76       0.75       0.75       0.75         .81       0.80       0.79       0.77       0.77       0.76       0.76       0.75       0.75       0.75         .81       0.80       0.79       0.77       0.77       0.76       0.76       0.75       0.75       0.75	.81 0.80 0.79 0.78	U.77	0.77	0.76	0.76	0.75	0.75	0.74	0.74
.81       0.80       0.77       0.77       0.77       0.76       0.75       0.75       0.75       0.75       0.75         .81       0.80       0.77       0.77       0.77       0.77       0.76       0.75       0.75       0.75       0.75         .81       0.80       0.79       0.77       0.77       0.77       0.76       0.75       0.75       0.75	.3ī 0.30 0.79 C.78	J.7E	0.77	0.76	0.76	0.75	0.75	0.75	0.74
.81 0.80 0.75 0.77 0.77 0.77 0.76 0.76 0.75 0.75 0.75 0.75 .81 0.80 0.80 0.79 0.77 0.77 0.77 0.76 0.76 0.75 0.75 0.75	.81 0.80 0.79 0.79	0.78	C.77	0.77	0.76	0.76	0.75	0.75	<b>0.</b> 75
.81 0.80 0.30 0.79 0.75 0.77 0.77 0.76 0.76 0.75 0.75 0.75	.81 0.80 0.79 0.79	0 <b>.</b> 70	0.77	0.77	0.76	0.76	0.75	0.75	0.75
	.81 0.30 0.30 0.79	32°0	0.77	0.77	0.76	0.76	0.76	0.75	0.75

1

	-
	-
(B)	+
Ч	17
ole	-
Та	•
	•
	•
	•

Variation of coefficient'b with temperature and salinity

<i>`</i>			en run and der Menten son				T °C					
	5		4		10	13	16	19	22	25	28	31
15 +	0,04	ပ. ပ)	-0,05	60"0~	<b>-</b> C.13	-0.17	-0.20	-0.23	-0.26	<b>-</b> 0.29	-0.31	-0.33
20 +(	0.03	<b>-</b> 0°05	-0.06	0[-0-	-0°14	-0,13	-0.21	<b>-</b> 0 <b>.</b> 24	-0.27	<b>-</b> C.29	-0.31	<b>-0.</b> 33
25 +(	0.01	-C, C4	-C. ÚS	-0.12	9 - 0	-0.19	-0.52	-0.25	-0.27	-0.30	<b>-0.</b> 32	-0.34
30 <b>+</b> (	0,01	-0° 02	<b>-</b> 0,03	<b>-0,</b> ]3		0;;0-	<b>-</b> 0,23	-0,26	<b>-</b> 0.28	<b>-</b> 0 <b>.</b> 30	-0.32	-0.34
35(	0, 02	-0°07	-0°10	15 12	8- - -	0,21	-0.24	-0.26	-0.29	<b>-0.</b> 31	<b>-</b> 0 <b>.</b> 33	<b>-</b> 0.34
40	0.04	-0.08	-0.12	S.⊒.•O+	óľ °Om	-0, 22 -0	-0.25	-0.27	-0.29	-0.31	<b>-</b> 0 <b>.</b> 33	-0.35
			and and and and and and and	a manging ang bang bang bang bang bang bang ban								

that the magnitude of the error in the computed value of  $\sigma_t$  is such that d  $\sigma_t$  may be taken as  $\pm 0.02$  when the salinity is determined by the method of conductivity ratio measurement and as  $\pm 0.03$  when the salinity is determined by the method of chlorinity titration.

This shows that the result can be in error by  $\pm 0.02$  in  $\sigma_{\pm}$  even when using the best available methods for the measurement of temperature and salinity. For more accurate results we may have to go for instrumental methods. One such instrument described by Cox et al. (1970), eventhough gives accurate results (the accuracy claimed is  $\pm 0.008$  in  $\sigma_{\pm}$ ), is not suitable for routine work because of its complexity. Kratky et al. (1969) has developed a more accurate density comparison instrument suitable for routine work. Although this instrument appears to be rather slower in operation compared to modern salinometers, it gives better results compared to computational methods. The accuracy claimed for the instrument is  $\pm 1_{*}5 \times 10^{-6}$  in density for a variation of temperature by <u>+</u>0.Cl<sup>0</sup>C. Assuming the temperature variation in the instrument to be within  $\pm 0.01^{\circ}$ C, since the maximum error in the determination of temperature is  $\pm 0.02^{\circ}$ C, the maximum error in the determination of

-71-

density in situ will be 4.5 x  $10^{-6}$  or less than half a unit in the second decimal place of the  $\sigma_t$  value. This means that, if the accuracy claimed for the instrument can really be achieved in the routine work, the result will be at least four times more accurate than the result obtained from computational methods.

# 4.2. Specific volume and specific volume anomaly

Instead of density in situ, its reciprocal,  $\alpha_{s,t,p}$ , called the specific volume in situ, is generally used in Dynamic Oceanography. In order to avoid writing a large number of decimals, the specific volume is commonly expressed as an anomaly,  $\delta$  defined as

$$\delta = \alpha_{s,t,p} - \alpha_{35,0,p}$$
 (4.15)

where  $\alpha_{35,0,p}$  is the specific volume of a 'standard ocean' of salinity 35%, at 0°C and at pressure p expressed in decibars. The values of  $\alpha_{35,0,p}$  for the 'standard ocean' were first tabulated by Bjerknes and Sandstrom (1910). The value of the anomaly  $\delta$ , which contains the departures of the real ocean from the 'standard ocean' depends on the temperature, salinity and pressure and hence can be expressed as

 $\delta = \delta_{s} + \delta_{t} + \delta_{s,t} + \delta_{s,p} + \delta_{t,p} + \delta_{s,t,p}$ (4.16) The last term of the above expression,  $\delta_{s,t,p}$ , is so small that it can always be neglected. The first three terms, which are independent of pressure, are usually grouped together and the sum is known as the thermo-steric anomaly,  $\mathcal{S}_{\mathrm{T}}$ . The value of  $\mathcal{S}_{\mathrm{T}}$  may be obtained from the value of  $\sigma_{\mathrm{t}}$  using the relation

$$\delta_{\rm T} = 0.02736 - \frac{10^{-3} \sigma_{\rm t}}{1+10^{-3} \sigma_{\rm t}}$$
(4.17)

Hence

$$\mathcal{S} = \mathcal{S}_{T} + \mathcal{S}_{s,p} + \mathcal{S}_{t,p} \tag{4.18}$$

The values of the three terms on the right hand side of the above expression may be obtained from Oceanographic Tables and hence the specific volume anomaly <u>in situ</u> may be computed. Sverdrup (1933) was the first to compile these tables.

The error in the result of computation for  $\mathcal{S}$  due to the errors in the measurements of temperature, salinity and pressure may be obtained by taking the total differential of the above expression.

$$d\boldsymbol{\delta} = d\boldsymbol{\delta}_{f} + d\boldsymbol{\delta}_{s,p} + d\boldsymbol{\delta}_{t,p}$$
(4.19)

The magnitudes of the errors  $d\delta_{s,p}$  and  $d\delta_{t,p}$  are small compared to the magnitude of error  $d\delta_{\Gamma}$  so that the former two are neglected in the following discussions.

Hence

$$d\delta \simeq d\delta_{T}$$
 (4.20)

 $d\mathcal{S}_{T}$  may easily be obtained from the relation (4.17) as

$$d \partial_{T} = -\frac{10^{-3} d \sigma_{t}}{(1+10^{-3} \sigma_{t})^{2}}$$
(4.21)

Hence

$$d d' \simeq d d_{T} = - \frac{10^{-3} \sigma_{t}}{(1+10^{-3} \sigma_{t})^{2}}$$
 (4.22)

The negative sign in the above expression is immaterial since  $d\sigma_+$  is just as likely to be positive as negative.

We have elready discussed the magnitude of  $d\sigma_{\pm}$ and found that it is  $\pm 0.02$  in  $\sigma_{t}$  when salinity is obtained using the method of conductivity ratio measurement and is  $\pm 0.03$  in  $\sigma_{\pm}$ , when the salinity is measured using the method of chlorinity titration. Using these values for the error in  $\sigma_{\pm}$  it can easily be estimated, using the expression (4.22), that the error in the specific volume anomaly is  $\pm 2 c l/ton$  when the former method is used for the determination of salinity and is  $\pm 3 c l/ton$  when the latter method is used for the purpose<sup>\*</sup>. Since the error in the determination of  $\sigma_t$  by

\*There is a widely used graphical method described by Montgomery (1954) and Montgomery and Wooster (1954) for

instrumental method, more specifically, when the instrument developed by Kratky <u>et al</u>. (1969) is used for the purpose, is less than half a unit in its second decimal place, the corresponding error in the value of

#### 

the determination of the values of thermosteric anomaly at different depths from known values of temperature and salinity. The method involves, as a first step, drawing of the temperature-depth curve for a station on a graph sheet which has the temperature marked along the abscissa, depth and salinity marked along the ordinate and super imposed with a family of curves representing thermosteric anomaly. Salinity is then plotted against temperature (This is essentially a f-S curve). Using these two curves, depths of chosen values of thermosteric anomalies are obtained.

The smallest division on the above mentioned graph sheet is equivalent to  $0.05^{\circ}$ C in temperature and 0.05/. in salinity and so plotting of a point on the T-S curve introduces an additional error of  $\pm 0.025^{\circ}$ C in temperature and  $\pm 0.025$ /. in salinity. Consequently the total error in the values of temperature and salinity, to be used for the purpose of computing the error in the value of thermosteric anomaly obtained from the station curves, specific volume anomaly will be less than half cl/ton.

should be taken as  $\pm 0.045^{\circ}$ C and  $\pm 0.045\%$ . respectively, assuming that salinity is obtained using the method of conductivity ratio measurement. This results in an error of approximately  $\pm 4$  Cl/ton in the value of thermosteric anomaly obtained graphically.

#### CHAPTER V

# ERRORS IN THE COMPUTATION OF DERIVED QUANTITIES

In chapter IV the magnitudes of errors that are caused in the computation of dependent oceanographic variables due to the random errors in the measurements of the independent oceanographic variables were discussed. These dependent oceanographic variables are used in the computations of the derived quantities, namely, the dynamic depth anomalies of isobaric surfaces and the relative currents. Since the dependent oceanographic variables are subject to errors, the derived quantities computed from them are also subject to errors, the magnitudes of which are discussed in this chapter.

### 5.1. Dynamic depth anomaly of isobaric surfaces

Level surfaces are surfaces that are everywhere normal to the force of gravity. Since potential energy of a mass remains constant on such a surface, they are surfaces of constant gravitational potential or constant geopotential. The ideal sea surface is a level surface with zero geopotential value assigned to it. The geopotential values of other equipotential surfaces are obtained by calculating the amount of work, W, required to move a unit mass from the ideal sea surface a geometrical distance h along a plumb line

$$i.e., \qquad \mathbb{N} = gh \qquad (5.1)$$

where g is the acceleration due to gravity.

The numerical value of the geopotential depends on the units used in gh. Bjerknes (1910) introduced, as a unit of the geopotential, the dynamic decimetre which represents the work that must be done to lift a unit mass along the plumb line about one geometric decimetre. The practical unit is the dynamic metre, D, and is defined by

$$D(dyn:m.) = \frac{gh}{10} - (\frac{m}{\sec c})^2$$
 (5.2)

where h is expressed in metres and g in metre/sec<sup>2</sup>. Since the numerical value of g, when expressed in m/sec<sup>2</sup>, is less than 10, the dynamic metre is slightly greatersmaller numerically than the geometric depth of one metre. A level surface in the ocean can then be defined as a surface of equal dynamic depth below the ideal sea surface.

The static pressure P at any depth h metres below the sea surface is given by the weight of the water column of unit cross-section between the sea surface and depth h. If  $\vec{\rho}$  is the mean density of the water column, then  $P = \vec{\rho}$  gh (5.3)

Substituting for gh from equation (5.2),

$$P = 10 \vec{P} \, \textit{D} \, \text{bar} = \vec{P} \, \textit{D} \, \text{decibar}$$
(5.4)

Hence dynamic depth expressed in dynamic metres is given by

where P is expressed in decibars and  $\overline{\alpha}$  is the mean specific volume of the water column. Since the density and the specific volume of the waters in the ocean vary in both the horizontal and vertical directions, accurate computation of the dynamic depth requires the infenitesimal form of the equation (5.5).

$$dD = \overline{\alpha} \, dp \tag{5.6}$$

The dynamic distance between two isobaric surfaces with pressures  $P_0$  and P is obtained from equation (5.6) as

$$\dot{\mathbf{D}} = \int_{P_0}^{P} \vec{\alpha} \, \mathrm{dP} \tag{5.7}$$

If  $P_o$  refers to the pressure at the sea surface, then  $P_o = 0$  and  $\cup$  represents dynamic depth of an isobaric surface with pressure P. Using equation (4.15), equation (5.7) may be written as

$$D = \int_{P_0}^{P} \alpha_{35,0,P} \, dp + \int_{P_0}^{P} \vec{\delta} \, dp$$
  
=  $D_{35,0,P} + \Delta D.$  (5.8)

The first term on the right hand side of the equation (5.8), namely,

$$D_{35,0,P} = \int_{P_0}^{P} \alpha_{35,0,P} \, dp \tag{5.9}$$

contains the contribution of a standard ocean which is invariable. The second term, namely,

$$\Delta D = \int_{P_0}^{P} \vec{\delta} dp$$
 (5.10)

contains the departures of the real ocean from the standard ocean and is called the dynamic depth anomaly between the isobaric surfaces  $P_0$  and P. Since only  $\Delta D$  is variable between two isobaric surfaces with pressures  $P_0$  and P, the relative geopotential intervals between the isobaric surfaces can be obtained from an evaluation of  $\Delta D$ , if  $\delta$ , is known as a function of depth.

Equation (5.10) may be used in the construction of dynamic topographies of given isobaric surfaces to provide

maps showing lines of equal dynamic depth anomaly of an isobaric surface below the sea surface. The dynamic topographies thus obtained are known as relative topographies since they are obtained with reference to the sea surface. Since the actual topography of the sea surface more or less deviates from a level surface, to get the absolute topography of the different isobaric surfaces we should select, as a reference surface, an isobaric surface which coincides with a level surface. A reference surface of this kind is called a reference level and can usually be found only deep below the ocean surface. Since the isobaric surface and the level surface coincide at a reference level, no water motion is possible at that level, and hence, may be called a level of no motion. When such a reference level is selected, the absolute topographies of the different isobaric surfaces are obtained by computing the dynamic height anomalies of the different isoparic surfaces taking this reference level as the zero reference surface. For this computation, equation (5.10) may be modified by changing the origin of the co-ordinate system from the sea surface to the reference level and assuming the Z axis to be directed vertically upwards, as

-81-

$$\Delta D = -\int_{P_0}^{P} \vec{\sigma} \, dp = -\int_{\sigma}^{P-P_c} \vec{\sigma} \, dp$$
$$= \int_{\sigma}^{P_c} \vec{\sigma} \, dp \qquad (5.11)$$

where  $P_0$  is the pressure at the reference level and P is the pressure at any other isobaric surface.

In practice, metres of depth are substituted for decibars of pressure in equation (5.11). This is permissible because the error introduced due to this substitution will be negligibly small, particularly in the lower latitudes and in the upper layers of the ocean, as we have already seen in section 3.2.

$$\therefore \Delta = \int_{0}^{H_{0}-h} \overline{\sigma} \, dz$$

$$= \int_{0}^{Z} \overline{\sigma} \, dz \qquad (5.12)$$

where H<sub>o</sub> and h are depths from the sea surface to the reference level and any other isobaric surface respectively and Z is the height of any isobaric surface above the reference level. Equation (5.12), in finite difference form, useful for computations with discrete data, is

$$\Delta D = \sum_{i}^{n} \bar{\sigma}_{i} \Delta \pi_{i}$$
(5.13)

-82-

where n is the number of intervals into which the distance between the level surface and the given isobaric surface is divided.

The error in the computation of dynamic height anomaly due to the errors inherent in the determination of specific volume anomaly and depth may be obtained by taking the total differential of equation (5.13).

$$d\Delta D = \sum_{i}^{n} \Delta z_{i} d\overline{\delta}_{i} + \sum_{i}^{n} \overline{\delta}_{i} d\Delta z_{i}$$
(5.14)

The contribution of the second term on the right hand side of the expression (5.14) to the total error in the determination of dynamic height anomaly may be estimated as follows. If we assume that  $\Delta z_i$  are error free depth increments, then an error in the determination of depth will appear as a change in the value of  $\overline{s}_i$ . We have earlier, in section 3.2 estimated the magnitude of error in the determination of depth as  $\pm 2$  m upto a depth of 200 m and  $\pm 4$  m below 200 m. The change in value of  $\overline{s}_i$ , for such depth differences, is negligible except where the variation of specific volume anomaly with depth is large. Sufficiently large variation which can introduce an appreciable change in  $\overline{s}_i$  for depth differences equal to the magnitude of error in the determination of depth is found only in the thermocline region, that is roughly in the depth range from 50 m to 350 m. If the selected reference level is sufficiently deep, the error contributed by this term to the total error in the computation of dynamic height anomaly will be small comp red to that contributed by the first term and so, this may be neglected. Hence expression (5.14) becomes

$$d\mathbf{\Delta} D = \sum_{i}^{n} \Delta z_{i} d\mathbf{\vec{d}}_{i}$$
$$= \sum_{i}^{n} \Delta z_{i} d\mathbf{\vec{d}}_{i}$$
(5.15)

 $\tilde{d}_i = \delta_i$  because  $\tilde{d}_i$  is obtained, in practical computations, from just two values of specific volume anomalies, found at the two extremities of the depth interval  $\Delta z_i$ .

Expression (5.15) shows that the error in the determination of dynamic height anomaly is the result of the sum of a large number of individual errors. Since the errors are just as likely to be positive as negative, their sum will never be large and the result should be obtained using the normal law of errors. Since we know only the maximum value of each individual error, we should use the formula (2.24) derived in section 2.3, which is given as

$$\sigma = \frac{\sqrt{s} \lambda e}{\sqrt{3}}$$

When the depth intervals  $\Delta z_i$  are constant and since we have assumed that  $d\delta_i$  is constant throughout the water column

$$\mathcal{L} \in = \Delta z \ d \delta \tag{5.16}$$
Hence expression  $(5.15)$  may be written as

$$\sigma = \frac{\sqrt{S}\Delta z \, d\delta}{\sqrt{3}} \tag{5.17}$$

Normally the total depth is divided into several depth ranges, each of which is divided into equal depth intervals. In such situations we should calculate  $\sigma$  for each depth range using formula (5.17) and then get the standard deviation of the combined error distribution which represents the distribution of error in the determination of dynamic height anomaly. This can be obtained using formula (2.22) of section 2.3, given as

$$\sigma = \sqrt{\sigma_1^2 + \sigma_2^2 + \cdots}$$

where  $\sigma_1$ ,  $\sigma_2$ , etc. are standard deviations for the different depth ranges. Twice the value of the standard deviation of the combined distribution may be taken as the maximum error in the determination of dynamic height anomaly, since more than 95%. of the time the error will be within the range of  $\pm 2 \sigma$  (Section 2.3).

Formula (5.17) shows that  $\sigma$  increases as S increases keeping  $\Delta z$  constant. This means that deeper we select the reference level, the more will be the error in the computed dynamic height anomaly. Also the formula shows that the larger the number of depth intervals into which a specific range of depth is divided, the smaller will be the computational error.

To illustrate computation of error in the dynamic height anomaly of an isobaric surface, let us compute the same for the zero decibar surface with reference to 1500 decibar surface. If we assume that data are available for the standard depths accepted by the Jational Oceanographic Data Centre, then there are three depth intervals of 10 m each in the depth range from 0 m to 30 m, one depth interval of 20 m in the depth range from from 30 m to 50 m, four depth intervals of 25 m each in the depth range from 50 m to 150 m, three depth intervals of 50 m each in the depth range from 100 m and twelve depth intervals of 100 m each in the depth range from 300 to 1500 m. The value of the maximum error,  $\ell \xi$ , for the 10 m depth interval is obtained using formula (5.16) as

> $\ell \xi = \Delta z \ d \delta$ = 10 decibar x 2 cl/ton (Section 4.2)

-86-

$$= 10 \times 10^{5} \frac{\text{gm cm}}{\text{sec}^{2} \times \text{cm}^{2}} \times \frac{2 \times 10 \times \text{cm}^{3}}{10^{5} \text{ gm}}$$
$$= 20 \frac{\text{cm}^{2}}{\text{sec}^{2}} = \frac{20}{10^{4}} \frac{\text{m}^{2}}{\text{sec}^{2}}$$
$$= \frac{20}{10^{5}} \text{ dyn. m} = 0.2 \text{ dyn. mm.}$$

Now the standard deviation  $\sigma$  for the depth range from O m to 30 m is obtained using formula (5.17) as

$$\sigma = \frac{\sqrt{S} \Delta z}{\sqrt{3}} \frac{d}{\sqrt{3}} = \frac{\sqrt{3} \times 0.2}{\sqrt{3}}$$
$$= 0.2 \text{ dyn. mm.}$$

Similarly standard deviation for the other depth ranges are computed after calculating the corresponding maximum errors and the results are tabulated in Table V. Using the values of the standard deviation for different depth ranges given in Table V, the standard deviation of the combined error distribution is obtained using formula (2.22) of section 2.3 as

$$\sigma = \sqrt[7]{\sigma_1^2 + \sigma_1^2 + \sigma_3^2 + \sigma_4^2 + \sigma_5^2}$$
  
=  $\sqrt[7]{(0.2)^2 + (0.2)^2 + (0.6)^2 + (1.0)^2 + (4.0)^2}$   
=  $\sqrt{0.04 + 0.04 + 0.36 + 1 + 16}$   
=  $\sqrt{17.44} = 4.2$  dyn. ma.

## Table V

Standard deviation of maximum error for different

# depth ranges

Depth range (m)	Depth interval (m)	Maximum error (dyn. mm)	σ for the depth range (dyn. mm)
0- 30	10	0.2	0.2
<b>30-</b> 50	20	0.4	0 <b>.</b> 2
50-150	25	0.5	0.6
150-300	50	1.0	1.0
300-1500	100	2.0	4.0

Table VI

Values of  $d\Delta D$  for different isobaric surfaces

with reference to 1500 d bar surface

						_ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~			3				
Jepth (m)	0	10	20	30	50	75	100	1 <b>2</b> 5	150	200	250	300	
d∠D (dyn.mm	)8	8	8 100 100 cm m	8	8 8	8. 8.	ප ප සංක ක	8	8		8	8	
Depth (m)	400	500	600	700	300	900	100	011	100	1200	1300	1400	15
d <b>A D</b> (Jyn.am	) <sup>8</sup>	7	7	7	б	6		5	5	4	3	2	
and one with the sub-		to read which have been		1. AND FUEL C. 1. C.	A 100 100 100 100 100 100	with							

The maximum error in the computation of dynamic height anomaly of the zero isobaric surface with reference to 1500 d bar surface is twice the above value of  $\sigma$  and so

 $d\Delta D = 2 \times 4.2 = 8.4 \implies 8 \, dyn. \, mm_{e}$ 

corrected to the nearest whole number. In a similar manner the maximum errors in the computation of dynamic height anomalies of the other isobaric surfaces with reference to 1500 d bar surface are computed and are presented in Table VI.

# 5.2. Aelative current

Relative current perpendicular to the vertical plane between two stations A and B is computed using Helland-Hansen formula (Sandstrom and Helland-Hansen, 1903), derived from Bjerkne's circulation theorem (Bjerknes 1393, 1900) with the assumptions that motion is non-accelerated and frictionless and that the observations are taken simultaneously. Though not absolutely justified, these assumptions are used for the computation of currents in the ocean, since the oceanic condition is generally quasi-permanent.

Helland-Hansen's formula is derived by equating the pressure gradient force acting down slope of a

sloping isobaric surface with the Coriolis force acting upslope. Solving for V, we get

$$V = \frac{10 (\Delta D_A - \Delta D_3)}{L. 2 \omega \sin \phi}$$
(5.13)

where

The current is called relative current because the current computed using the formula is relative to any unknown current at the reference level. To get the absolute current, the reference level selected should be a level of no motion.

The error in the computation of relative current is due to the errors in the computed value of difference in dynamic height anomalies, in the measurements of distance between stations and in the latitude angle. This may be obtained by taking the total differential of the expression (5.18).

$$dV = \frac{10d (\Delta D_{A} - \Delta D_{B})}{L \cdot 2 \omega \sin \phi} - \frac{10 (\Delta D_{A} - \Delta D_{B})}{L^{2} \cdot 2 \omega \sin \phi} dL$$
$$- \frac{10 (\Delta D_{A} - \Delta D_{B})}{L \cdot 2 \omega \sin^{2} \phi} \cos \phi d\phi \qquad (5.19)$$

The second and third terms on the right hand side of the equation (5.19) may be discarded being negligible compared to the first term assuming that the measurements of distance between stations, L and latitude angle, $\phi$  are done accurately. This is generally true, particularly with the use of modern mavigational aids. Hence the error in the relative current may be written as

$$dV = \frac{1 \cup d (\Delta D_A - \Delta D_B)}{L \cdot 2 \omega \sin \phi}$$
(5.20)

Expression (5.20) shows that the error in the computed current is directly proportional to the error in the difference in dynamic height anomaly and is inversely proportional to the distance between stations and the sine of the latitude angle<sup>\*</sup>. To illustrate the computation of error in the relative current relative to 1500 d bar surface, let us assume that the distance between the two stations A and B is 100 kilometres and the average latitude angle is  $5^{\circ}$ . The maximum error in the difference between the dynamic height anomalies at the two stations should be computed as twice the standard deviation of the combined distribution resulting from the combination of the

equation (5.18) will lead to computation of infinite current at the equator. But the fact that halving the distance between stations will also lead to unrealistic results, by doubling the error in the computed current, is not recognised by many oceanographers. Montgomery and Stroup (1962) found that decrease of station spacing will not always result in more details of the current distribution and stated: 'In effect, the stations (9,11,...., 27) midway between whole degrees of latitude are neglected. It was thought that the details gained by halving the spacing of verticals would be largely unreal'. Again while discussing the representation of distribution of geostrophic flow through a vertical section they stated: 'It might be thought that a continuous distribution of geostrophic velocity component could be attained by sufficiently

two error distributions of the dynamic height anomalies at the two stations: Hence

d 
$$(\Delta D_A - \Delta D_B) = 2\sqrt{42 + 4^2} = 2\sqrt{2} \times 4 = 11.3$$
  
 $\implies 11 \, dyn. \, ma$ 

The error in the computed current, dV, is computed from

reducing the station spacing. The result, however, would be an increasingly irregular pattern, ultimately bearing no resemblence to the actual current (because with decreasing spacing even small fluctuations of the observed specific volume would lead to increasingly swift and narrow current bands of alternating direction)'. One of the reasons for the above results obtained by Montgomery and Stroup (1962), even though they have taken a comparatively shallow reference surface at 300 m which should have resulted in less computational error, may be the increased error inherent in the method of obtaining values of thermosteric anomaly described by Hontgomery (1954) and Montgomery and Moster (1954) (Section 4.2). Ofcourse, smaller spacing of verticals will result in gained details for the vertical sections of the dependent and independent parameters, but not of the distribution of currents.

the formula (5.20) and using the LaFond's tables (LaFond, 1951) as

dV = 9 cm/sec

Figure 2 shows the variation of the error in the computed current (relative to 1500 d bar surface) with latitude angle for an error of 10 dyn. mm in the difference in dynamic height anomaly and for a pair of stations separated by 100 km. The curve shows that the error is less than 1 cm/sec above 40° latitude angle and increases towards the equator. Very near the equator the rate of increase is very large.

Any computed current which is less than or equal to the magnitude of the error itself should be considered as unreal. Since the error is given in absolute magnitude, its effect is serious where the current velocities are small, particularly very near the equator where the magnitude of error increases due to latitude effect.

-93-

for 10  $\mathrm{dyn}_\bullet\mathrm{m}\mathrm{m}_\bullet$  error in the difference in dynamic height anomaly and for 100 Km station spacing (new trive to issuch that)


#### CHAPTER VI

## ILENTIFICATION OF A ZERO MEFERENCE LIVEL

To get the absolute topography of the different isobaric surfaces, an isobaric surface which coincides with a level surface should be selected as a zero reference level. Since no water notion is possible at this level, it is called a level of no motion. Proper selection of a zero reference level is important for the presentation of an oceanic circulation pattern. This chapter presents a method for the identification of a zero reference level.

### 6.1. Determination or an appropriate reference surface

For the determination of the absolute magnitudes of computed currents, the reference surface used for the computation of the relative currents should either be a level of no motion or the distribution of the current velocity along the reference surface should be known. The latter method is not widely used for the following reasons:

 Current measurement at sea is very difficult and subject to uncertainties. 2) The measured current is the total current which may include the effects of winds, tides, internal waves, etc. in addition to the geostrophic component of the current. Hence, for the computation of absolute currents, identification of a 'level of no motion' is usually attempted.

Literature cites several methods for the identification of a level of no motion. Some earlier investigators assumed a level of no motion as deep as possible reasoning that the velocity decreases with depth and that at great depths the isobaric surfaces are nearly horizontal. This method has the disadvantage that the error in the computed current may make the picture of the circulation pattern completely unreliable since the error in the computed currents increases with increase in the depth of reference surface. Other methods developed by different authors were critically examined by Fomin (1964) and may be summarised as follows:

### a) <u>\_ietrich's method</u>

The idea that the intermediate oxygen minimum layer in the ocean corresponds to a layer with minimum horizontel water motion was first advanced by Jacobsen (1916) and was further developed and applied in practice

-95-

by Aust (1935) and Dietrich (1936). Several authors have refuted this idea (Rossby, 1936; Iselin, 1936; Mattenburg, 1938; Seiwell, 1937) and have argued that an oxygen minimum layer need not necessarily be a layer of minimum current velocity. Also Sverdrup (1938) showed that an oxygen minimum layer cannot have any general dynamic significance.

b) Parr's method

Parr (1938) advanced the idea that since the thickness of a layer bounded by two isopycnic surfaces (surfaces of equal ensity) cannot remain constant in the region of a current and should vary perpendicular to the direction of current, where the thickness of such a layer is constant, the horizontal water motion is zero or it is a layer of no motion. He also suggested a graphical method to identify such a layer. Fomin (1964) showed that the above is a necessary but not a sufficient condition for the existance of a layer of no motion because the vertical variation of current velocity depends not only on the slope of the isopycnic surfaces but also on the vertical density gradient. A layer bounded by two isopycnic surfaces in a region where there is a strong vertical density gradient will be least distorted compared to the overlaying and underlaying layers in the presence of a strong geostrophic current.

## c) Hidaka's method I

The current velocity field in the sea is in constant interaction with the field of any physical or chemical property of sea water and there is mutual adjustments between these two fields. This fact was used by Hidaka (1949) to develop a method for the determination of the level of no motion in the sea from the salinity distribution. He argued that the surface where the second derivative of the vertical salinity distribution,  $\frac{\partial^2 S}{\partial z^2}$ , is equal to zero should be a level of no motion. Fomin (1964) showed that Hidaka's method provides the identification of a surface that has very definite structural features of the salinity field, but that such features are not uniquely related to the current velocity field.

## d) Hidaka's method II

Hidaka (1940a,1940b,1950) suggested another method, for the identification of the level of no motion, based on continuity considerations for the stationary distribution of certain physical and chemical properties of sea water. He considered a volume of water in the form of a tetrahedral prism that extends from sea surface to the bottom. Assuming continuity of water volume, Hidaka obtained a set of equations, the solution of which lead to the identification of the level of no motion. Defant (1941a), however, has raised objection on the practical applicability of this method on two grounds:

i) The assumption on the continuity of water volume is not strictly true theoretically because, continuity condition requires constant mass and not constant volume.

ii) The set of equations obtained by Hidaka is practically inconsistent and cannot be solved with the existing accuracy of measurements at sea.

e) Sverdrup's method

Sverdrup <u>et al</u>. (1942) suggested a method for the identification of the level of no motion which uses the known fact that, in the steady state, the total water transport through any cross section of an oceanic area, which extends from one shore to the other, should be zero. He considered a horizontal reference surface and argued that the reference surface will be in the layer of no motion when the water transport through the section above the reference surface is equal to the water transport below the reference surface. Fomin (1964)

-98-

- G 3427-

-99-



points out that this method is unsuitable on threasand counts:

i) The currents that compensate each other need not necessarily be in the vertical plane. They can be in the horizontal plane as well.

ii) The accuracy of water density determination is insufficient for a successful computation of water transport, particularly in the deep layers.

iii) For a successful identification of the level of no motion, the pure drift component of the current should also be considered which will not only complicate the computation but also introduce additional errors caused by the uncertainty of the wind field that corresponds to the stationary case.

f) Defant's method

Defant (1941b), while comparing the differences in the relative dynamic depth anomalies of given isobaric surfaces between adjacent oceanographic stations in the Atlantic ocean, found that at certain levels these differences were practically constant over a large depth interval. Such constant relative pressure differences between adjacent stations can be interpreted in two ways: i) The whole layer of deep water has a constant velocity.

ii) The whole layer is uniformly at rest.

The first interpretation should be considered as unreasonable since the constant velocity in most cases proves to be considerably larger than the surface velocity, a result which is against the present oceanographic experience. Hence, Pefant concluded that the second interpretation is valid and the layer may be considered as a layer of no motion.

Eventhough Defant's method is the most justified and widely used one for the identification of the layer of no motion, it is not without objections. *A*ust (1951) put forward two arguments against the method:

i) It is doubtful whether we can find a level of no motion in a layer of constant relative pressure differences when the current computed has the same direction above and below the layer.

ii) If current is computed in the immediate vicinity of the bottom relative to a level of no motion identified in a layer of constant relative pressure differences, it is frequently found that the magnitude of the computed

-100-

current exceeds the magnitude of the current at the surface and in a direction opposite to the direction of the surface current. This is an unreasonable result according to the present knowledge of the oceans.

Fomin (1964) pointed out that in a vertical section of the differences in the relative dynamic depth anomalies of isobaric surfaces between adjacent stations, selection of the depth of level of no motion is not always unique because, in some cases, there may be several layers with constant relative pressure differences and selection of any one of them is arbitrary. Also, for some pairs of stations, there may not be a layer with constant relative pressure difference at all. He also pointed out that in a region where the current velocity is low, Defant's method is liable to fail because the relative dynamic depth differences between adjacent stations in such regions will be of the same order of magnitude as the computational error itself. The method is liable to fail in a weakly stratified water body for the same reason. It will give good results only in strongly stratified bodies of water and in regions where the current velocity is high.

Sastry and D'Souza (1971) have reported that application of Defant's method in the Arabian sea region

-101-

did not yield a zero level which could be accepted with any reasonable degree of certainty.

## g) Mamaev's method

Mamaev (1955) suggested a method which is very similar to Defant's method. His method is based on the fact that in a layer where the differences in relative dynamic depth anomaly between adjacent stations is constant, the specific volume anomalies should be equal. Hence a vertical distribution of the differences in specific volume anomalies at two neighbouring stations will show a zero value whenever the differences in the relative dynamic depth anomalies between these two stations are constant. This method has the advantage that the minimum or zero value of the differences in specific volume anomaly is easily found compared to a layer where the differences in dynamic depth anomalies are constant. Since this method is similar to Defant's method in principle, this method also is subject to the objections raised in the context of the latter.

# 6.2. Identification of a level where the current velocity is negligibly small.

The foregoing discussion on the methods used for the identification of the level of no motion in the sea

-102-

brings out the fact that there is no fool-proof method for the same. In a complicated mass field, as in the real ocean, it is unreasonable to think either that the level of no motion coincides with an isobaric surface over large areas or that it coincides with real surfaces in the ocean such as isothermal surface, isohaline surface, etc. In oceanic areas where the current continues from the surface right to the bottom without changing direction, a level of no motion will not exist at all. A surface where there is no water motion should be considered as having a very complex topography, which sometimes crop out at the surface where there are two opposing currents in the horizontal plane. The identification of such a surface in an oceanic area is beyond the means of present day dynamic methods in oceanography. Hence it is advisable to try to identify a level where the water motion is negligibly small and use this level for the computation of currents so that the computed currents will not be very much different from the absolute currents. A method for the identification of such a level has been described by Fomin (1964) by the application of the density model of Shtokman (1950,1951) to the oceanic mass field.

-103-

The component of the geostrophic current velocity,  $V_x$ , in the northern hemisphere, perpendicular to a vertical cross section is obtained by equating the pressure gradient force with the Coriolis force, assuming that the motion is non-accelerated and frictionless.

i.e., 
$$f V_{X} \rho = \frac{\partial p}{\partial y}$$
 (6.1)

where  $f = 2 \omega \sin \phi$  is the Coriolis parameter,  $\rho$  is the density of water and p is the sea pressure. The right hand side of equation (6.1) may be written in terms of density  $\rho$  by the use of hydrostatic equation, namely,

$$dp = g \rho dz \tag{6.2}$$

where g is the acceleration due to gravity and z is the depth from the sea surface. Integrating Equation (5.2) we get

$$\int_{0}^{z} dp = p = g \int_{0}^{z} \rho dz$$
 (6.3)

Differentiating equation (6.3) with respect to y

$$\frac{\partial p}{\partial \gamma} = g \int_{0}^{z} \frac{\partial P}{\partial y} dz$$
(6.4)

Substituting equation (6.4) in equation (6.1)

$$f V_x \vec{\rho} = g \int_0^z \frac{\partial f}{\partial y} dz$$

$$V_{\rm X} = \frac{g}{f \rho} \int_{0}^{z} \frac{\partial \rho}{\partial y} dz \qquad (6.5)$$

where  $\overline{\rho}$  is the average density of the water column.

Shtokman's density model, when applied to the oceanic mass field, assumes that the density at the reference surface, which is the lower boundary of the baroclinic layer, is constant and has a value equal to P(o) and that the deviation of density, P(x,y,z), at any point from this constant value is given as the product of two functions, one of which depends only on the vertical coordinate, and the other only on the horizontal coordinate.

i.e., 
$$P(0) - P(x,y,z) = \delta(z) f(x,y)$$
 (6.6)

where f(x,y) is known as the function of influence. Hence, along a vertical, f(x,y) should be a constant.

i.e., 
$$P(0) - P(z) = K \delta(z)$$
 (6.7)

The assumption that the function of influence is constant implies that, in an oceanic region where the vertical distribution curves of density are similar in appearance, the function O(z) will be different only by a constant multiplier. This will be true in oceanic areas where the T-S curves are similar and similarity of T-S curves is retained over very large oceanic areas. Now let us locate the origin of the coordinate system on the assumed reference surface with the z axis directed vertically upwards. Substituting equation (6.6) in equation (6.5) we get

$$V_{x}(z) = - \frac{g}{f \rho} \frac{\partial f(x, y)}{\partial y} \ll \int_{0}^{z} \delta(z) dz \qquad (5.8)$$

Assuming z = H at the sea surface, the geostrophic current at the sea surface is

$$V_{\mathbf{x}}(\mathbf{H}) = \frac{-g}{f \mathbf{\rho}} \frac{\partial f(\mathbf{x}, \mathbf{y})}{\partial \mathbf{y}} \stackrel{\mathbf{H}}{\kappa} \int_{\mathbf{0}}^{\mathbf{H}} \mathbf{d}z \qquad (3.9)$$

Solving for  $\frac{\partial f(x,y)}{\partial y}$  from equation (6.9) and substituting in equation (6.8) we obtain

$$V_{\mathbf{x}}(z) = V_{\mathbf{x}}(\mathbf{H}) \stackrel{\mathbf{0}}{\stackrel{\mathbf{0}}{=}} = V_{\mathbf{x}}(\mathbf{H}) \phi(z) \qquad (5.10)$$

where  $\phi$  (z) is known as the stratification function. Equation (6.10) describes the vertical distribution of geostrophic current velocity and may be used to compute any subsurface current, once the absolute value of the geostrophic component of the sea surface current is known by measurement. Equation (6.10) shows that the strarification function will be the same where the function  $\delta$ (z) at different points is different only by a constant multiplier. Integration of equation (5.10) between the limits z = 0 to z = H yields

$$\int_{0}^{H} V_{x}(z) dz = \int_{0}^{H} V_{x}(H) \psi(z) dz$$
$$= V_{x}(H) \int_{0}^{H} \psi(z) dz$$
$$= \frac{V_{x}(H)}{F(H)}$$
(6.11)

where the function F(H) is defined by

$$F(H) = \frac{1}{\int_{0}^{H} \phi(z) dz}$$

$$= \frac{0}{\int_{0}^{H} \int_{0}^{z} \phi(z) dz}$$
(6.12)

If the geostrophic current velocity component in equation (5.11) is considered as average current between two stations at a distance L, then the time rate of water transport between the verticals at the two stations between the sea surface and the assumed reference surface is given by

$$L \int_{0}^{H} V_{x}(z) dz = \frac{L V_{x}(H)}{F(H)}$$
(6.13)

It is easy to see that since L  $V_x(H)$  is a constant, the time rate of water transport varies inversely with the function F(H). It will be illustrated later in this section that function F(H) decreases in magnitude as the value of H increases, and becomes constant below a particular value of depth of the assumed reference surface. Hence the time rate of water transport remains constant irrespective of the reference surface selected, once it is below the particular reference surface where the value of the function F(A) becomes constant. This means that the layer of water below this reference surface is a layer of no motion. Hence the reference surface where the value of the function F(H) becomes constant is a level of no motion. The function F(A) will be the same in a region where the function  $\delta(z)$  is different only by a constant multiplier.

In fact, Shtokman's density model, when applied to the mass field of the ocean, does not ensure that the reference level thus identified is actually a level where the geostrophic current velocity is zero. The model only tells that if a current exists at the selected reference level, it will continue unabated to the bottom, because the constantly of the function F(H)in a layer indicates zero vertical density gradient, a condition which cannot change the magnitude of current in the vertical. The result that reference surface identified by the application of Shtokman's model to the oceanic mass field is a level of no motion is due to a tactical assumption that the velocity of geostrophic current at the sea bottom is zero. Since we generally expect only negligibly small current at the bottom of the sea where depths are large, the reference surface selected by the application of Shtokman's density model may be considered as a surface where the geostrophic current velocity is negligibly small and hence may be considered as a zero surface for the purpose of computing absolute currents.

Shtokman's density model cannot be applied in an oceanic region where the current changes in direction with depth and where it first increases and then decreases with depth. This is because of the assumption that the function of influence is constant along the vertical and so the velocity of geostrophic current decreases with depth without change in direction. Hence the boundary between two oppositely directed currents can only be vertical.

This method is applied to the Arabian sea region for the identification of a zero reference level where the

-109-

geostrophic current velocity is negligibly small. The data used are obtained from the Cruise No.31-0197 of the U.S. Research Vessel 'Atlantis' in the Arabian sea during August-September 1963. Fig.3 gives the location of the stations used in the present study.

We have already seen that Shtokman's density model can be applied only in an oceanic region where the magnitude of current decreases with depth and where the direction of current does not change with depth. This cannot be generally expected in the surface layers, particularly in a tropical oceanic region like the Arabian sea. So, for the purpose of computing function F(H) using Shtokman's density model, an oceanic region below about 1500 m is selected where generally the current velocity is expected only to decrease with depth with no change in direction. A region below 1400 m is taken for the purpose so that z = H now represents the 1400 d bar surface.

The method of computing function F(A) for Station No.82 is shown in Table VII assuming that the lower boundary of the current is 4000 m. For the puppose of this computation it is convenient to write the equation for function F(A) in finite difference form as Fig. 3: Map showing location of stations.



## Table VII

# Computation of function F(H)

H-z (m)	σ <sub>t</sub>	k <b>ó</b> (z)	κ <b>σ</b> (z)	κό(z)Δz	$I = K \sum_{0}^{z} \delta(z)$	)Δz ]	ĪΔz				
1	2	3	4	5	6	7	8				
1400	27.65	0.16	0.14	14 ()	64.0	<b>57</b> O	5700				
1500	27.69	0.12	0.14	14.0	50.0		5700				
1750	27.75	0.06	0.09	22.5	27.5	38.8	9700				
2000	27.78	0.03	0.05	12.5	15.0	21.3	5325				
2500	27.30	0.01	0.02	10.0	5.0	10.0	5000				
3000	27 81	0.00	0.01	5.0	0.0	2,5	1250				
4000	27.01	0.00	0.00	0.0	0.0	0.0	0.0				
4000	27.81	0.00			0.0						
$E_{\Delta z} = 26975$											
	$F(H) = 2.37 \times 10^{-3}$										
		= 2.	37 x 10	-51							

$$F(H) = \frac{0}{\int_{0}^{H} \int_{0}^{z} \delta(z) dz}$$

$$= \frac{1}{\sum_{0}^{H} \int_{0}^{z} \delta(z) (dz)^{2}}$$

$$= \frac{0}{\sum_{0}^{H} \int_{0}^{z} \delta(z) \Delta z}$$

$$(6.14)$$

In Table VII, the first column gives the depth of the different isobaric surfaces from the sea surface. The vertical distribution of  $\sigma_{\rm t}$  is given in the second column. The third column shows the values of  $[(\delta(z)],$  that is, values of  $[\sigma_{\pm}(o) - \sigma_{\pm}(z)]$ . The  $[K \delta(z)]$  values averaged by layers are given in the fourth column. In the fifth column are given the products of the average values of [K d(z)] and the corresponding depth intervals. The sixth column gives the summation  $\Sigma \int (z) \Delta z$  where z can have values in the interval  $0 \le z \le H$ . Column 7 contains values of the sixth column averaged by layers. The last column shows the average values of the seventh column multiplied by the corresponding depth intervals. Now the function F(H) is obtained by dividing the number in the first row of the sixth column, which represents  $(\tilde{z} \delta(z) \Delta z)$ , by the sum of the values of the last column,

-111-

which represents  $K \Sigma (\Sigma \overline{\mathfrak{G}(z)} \Delta z) \Delta z$ . To convert the result to the cgs system, we must multiply the same by  $10^{-2}$ . The same computations are repeated for the same station assuming the lower boundary of the current at 3000 m, 2500 m, 2000 m and 1750 m. Similarly function F(H) for different depths of the lower boundary of current are computed for the stations 48,88,103 and 111. The stations selected are well-spread over the whole of the Arabian sea region so that a zero surface identified with this method may be applied to the whole of this oceanic region. The computed values of function F(H)are tabulated in Table VIII. The first column shows the assumed depths of the lower boundary of the current from the sea surface and the other columns show the computed values of function F(H) for the different depths and for the selected stations. Fig.4 is a graphical representation of function F(H) for the stations 48,88 and 103. Both Table VIII and Fig.4 show that the function F(H) depends little on position and agree rather well in magnitude. This means that in the region selected the water is comparatively uniform in the horizontal plane. The I-S characteristics of the water masses of the Indian ocean published by Sastry (1971) and Sastry and D'Souza (1972) confirm this fact by showing that the T-S curves of the water masses of the Arabian sea region coincide

-112-

## Table VIII

Values of F(H) x  $10^{-5}$  cm<sup>-1</sup> for different depths of assumed reference surfaces

H-z	Station Numbers								
(m)	48	82	88	103	111				
1750	7.07	7.38	7.72	7.38	7.72				
2000	4.80	4,92	4.59	4.92	4.69				
2500	2.87	3.39	2.83	2.92	2,83				
<b>3</b> 000	2.25	2.37	2.24	2.27	2.18				
4000	2.25	2.37	2.24	2.27	1.53				



below 40 cl/ton isosteric surface which represents an oceanic region below 1500 m. The table and the figure also show that the function F(H) changes rapidly in the comparatively shallow regions and except for station 111 (Table VIII), the function F(H) is independent of depth when the depth is greater than 3000 m. At station 111 the vertical density gradient is not zero below 3000 m but is very weak compared to the upper layers. Hence, we may take the 3000 d bar surface as a zero surface in the Arabian sea with negligibly small geostrophic current.

#### CHAPTER VII

## CUMPUTATION OF DYNAMIC HEIGHT ANOMALY

In the preceding chapter we have found that 3000 d bar surface is an ideal surface to be used as a zero surface in the Arabian sea for the purpose of computation of the derived quantities, namely, anomalies of dynamic height of isobaric surfaces and geostrophic currents. Equation (5.17) has shown that the error in the dynamic height anomaly of an isobaric surface increases with increase in depth of the reference surface. It can further be shown, that the maximum error in the dynamic height anomaly of the zero isobaric surface with reference to 3000 d bar surface is 20 dyn. mm., two and a half times more than the maximum error calculated with 1500 d bar surface as a reference surface. This increased error in the computation of dynamic height anomaly of the sea surface will make the results on the surface circulation pattern more undependable. To reduce the error caused by the selection of a deep isobaric surface as a zero surface, a method is used in which the dynamic height anomaly of the different isobaric surfaces are computed using 1500 d bar surface as a reference surface and then the computed geostcophic current

velocities are reduced to the value relative to the 3000 d bar surface using Shtokman's density model. The 1500 d bar surface is selected for the following reasons:

i) 1500 d bar surface is a surface where the density variation in the horizontal is very small. The dynamic topography of the different isobaric surfaces presented with the 1500 d bar surface as a reference surface will not be very different, for practical purpose, from that presented with 3000 d bar surface as the reference surface.

ii) According to the NOUC accepted standard depths, there are frequent sampling points upto 1500 m depth which will help to reduce the error in the computed dynamic height anomaly.

iii) The mass field in the oceanic region below 1500 d bar surface is suitable for the application of Shtokman's density model so that the geostrophic currents computed with reference to 1500 d bar surface can be reduced to the value relative to 3000 d bar surface using this model.

# 7.1. Extending the selected zero reference surface into shallow regions.

Once the reference surface for the purpose of computation is selected, we must extend the computations

-115-

into shallow regions of the ocean where the depth is less than the depth of selected reference surface. Different methods available for this purpose were critically examined by Fomin (1964) which may be summarised as follows:

## 7.1.1. Helland-Hansen's method

In this method proposed by Helland-Hansen (1934), the block of solid earth in the form of a triangle formed by the reference line, the bottom line and the vertical at the shallow station is replaced by an imaginary water mass. If it is assumed that the gradient current velocity and the horizontal pressure gradient are zero along the bottom line, then the isobaric and isosteric surfaces in the imaginary water mass must be horizontal. This implies a motionless water mass. If the points of intersection of the isosteric lines with the bottom line are projected horizontally into the vertical at the shallow station, we get the vertical distribution of specific volume anomaly at the shallow station up to the reference surface and these values may be used for the computation of currents with reference to the selected zero surface.

The above method is based on the assumption that gradient current velocity and hence the horizontal pressure

-116-

gradient are zero near the bottom. According to Fomin (1954) this assumption is not justified because the velocity of total current at a solid boundary becomes zero due to friction and the consequent appearance of a compensating current. This compensating current cannot be determined by dynamic methods. Another disadvantage of the method is that its application requires graphical construction.

#### 7.1.2. Jacobsen and Jensen method

Jacobsen and Jensen (1926) proposed a method which is also based on the assumption that the current velocity and the horizontal pressure gradient along the bottom line are zero. Here also the solid earth formed by the reference line, the bottom line and the vertical at the shallow station is replaced by an imaginary motionless water mass. But this method has two more additional assumptions.

a) The bottom line between the two stations, one of which is shallow, is rectilinear.

b) The isosteres are equidistant near the bottom. These two assumptions result in the isosteres in the imaginary water mass being equidistant and parallel.

-117-

The practical computation is done in the following manner. Let A and B are two stations of which A is a shallow station and B a deep station through the bottom of which passes the zero reference surface which is horizontal. Now assume a horizontal reference surface passing through the bottom of the shallow station A and compute the differences in dynamic height anomalies between the two stations with reference to this assumed reference level. Then add to this the correction

$$\Delta = \frac{1}{2} \Delta p \left( \delta_{A_1} - \delta_{A_1} \right)$$
(7.1)

where  $\Delta p$  is the pressure difference between the assumed beference surface and the zero reference surface and  $\delta$  and  $\delta$  are the specific volume anomalies at the  $A_{A}$  assumed reference level for the two stations B and A respectively. The sum will then be the difference in dynamic height anomalies between the two stations with reference to the zero reference surface.

We have already seen in section 7.1.1. that the assumption of zero gradient current velocity and zero pressure gradient near the bottom is not justified. Eventhough this method has an advantage over the Helland-Hansen method in that its application does not require any graphical constructions, the additional

-118-

assumptions concerning the structure of the density field near the bottom is not justified because under actual conditions water density field rarely has such a structure (Fomin, 1964).

# 7.1.3. Goren's method

Goren (1948) proposed a method in which he suggested that the density field be extrapolated in the imaginary water mass in such a manner that all the isosteres on each horizontal level in the imaginary water mass have a constant slope equal to the slope of the isostere at that level on the bottom line. This construction will give the required vertical distribution of the specific volume anomaly at the shallow station with reference to the zero reference surface.

The advantage of this method over the two methods already discussed is that here the solid block of earth along the profile located above the zero reference surface is replaced with an imaginary moving baroclinic water mass. This eliminates the unjustified assumptions made in the former two methods that the geostrophic current velocity and the pressure gradient are zero near the bottom. But this method has also some short comings, the most objectionable one being the requirement that the slope

-119-

of all the isosteres should be constant at each horizontal in the imaginary water mass, a requirement very difficult to justify (Fomin, 1964). Further, the application of this method requires graphical construction.

Another method suggested by Fomin (1964) is th simple extrapolation of the specific volume anomaly of water along the profile in the imaginary water mass which will provide the vortical distribution of the specific volume anomaly at the shallow station up to the zero reference level. In this case also, solid block of earth along the profile located above the zero reference surface is replaced with a moving water mass. The subjective errors inherent in the method, however, make it most unsuitable, particularly if the shallow station is located at the boundary of the ocean under study.

# 7.2. A new method suggested for the extension of computation into shallow regions

The foregoing discussion on the different methods used for the extension of computation into the shallow regions of the ocean has brought out the fact that none of them are without short comings. Hence a new method is suggested for this purpose in which most of the objection raised in the preceding pages are avoided.

-120-

In this method, the dynamic relief blong a profile of the isobaric surface corresponding to the deepest sampling depth at the shallow station is linearly extrapolated along the profile in the shallow region. This gives the dynamic height anomaly of the isobaric surface at the shallow station with reference to the zero surface. If A,B and C are three stations in a line, of which C is a shallow station, then the required dynamic height anomaly at the shallow station is given by

$$\Delta D_{c}, = \begin{bmatrix} \Delta D_{A}, -\Delta D_{B}, \\ ---\overline{AB} \end{bmatrix} BC + \Delta D_{B}, \qquad (7.2)$$

where  $\Delta D_{A'}$ ,  $\Delta D_{B'}$ , and  $4 D_{C'}$ , are dynamic height anomalies of the isobaric surface passing through the deepest sampling depth at station C, and AB and BC are distances between stations A and B and stations B and C respectively.

The following are some of the advantages of this method:

a) In this method the section of the solid earth along the profile located above the zero reference surface is replaced with a moving baroclinic water mass.

b) Since it is a linear extrapolation, its application does not require graphical construction.

c) The linear extrapolation done only at one depth avoids accumulation of subjective errors.

If the values of the independent oceanographic parameters are missing at one depth at a particular station and if their values are known for the neighbouring stations on either side, the dynamic height anomaly at that depth may be obtained by linear interpolation. If  $A_1$ ,  $B_1$  and  $C_1$ are three stations in a line in that order and if the values of the independent oceanographic parameters are missing at some particular depth at station  $B_1$ , then the dynamic depth anomaly at the depth where the values are missing is obtained as

$$\Delta D_{A_{1}} = \begin{bmatrix} \Delta D_{C_{1}} \\ - \frac{\Delta D_{C_{1}}}{A_{1}C_{1}} \end{bmatrix} = \begin{bmatrix} \Delta D_{C_{1}} \\ - \frac{\Delta D_{C_{1}}}{A_{1}C_{1}} \end{bmatrix} = \begin{bmatrix} \Delta D_{C_{1}} \\ - \frac{\Delta D_{C_{1}}}{A_{1}C_{1}} \end{bmatrix} = \begin{bmatrix} \Delta D_{C_{1}} \\ - \frac{\Delta D_{C_{1}}}{A_{1}C_{1}} \end{bmatrix}$$
(7.3)

where  $\Delta D_{A_1}$ ,  $\Delta D_{B_1}$  and  $\Delta D_{C_1}$  are dynamic height anomalies of the isobaric surface passing through the depth at station  $B_1$  where the independent oceanographic parameters are missing and  $A_1C_1$  and  $B_1C_1$  are distances between station  $A_1$  and  $C_1$  and stations  $B_1$  and  $C_1$  respectively.

Assuming 1500 d bar surface as the reference surface and then extending the same into the shallow regions using the above method for the purpose of computation, the dynamic height anomalies of the different isobaric surfaces,

-122-

corresponding to the standard depths above 1500 d bar surface, were computed for stations along the profile approximately 15<sup>0</sup> latitude in the Arabian sea. The station positions are shown in Fig.3.

7.3. <u>Smoothening the dynamic relief in a profile of an</u> isobaric surface

In section 5.1 we have seen that the computed dynamic height anomaly of an isobaric surface will be in error by an amount which can be calculated using the method described in that section. Table VI of the same section gives the magnitudes of errors in the computed dynamic height anomalies of the different isobaric surfaces with reference to the 1500 d bar surface. The computed dynamic relief of an isobaric surface in a profile, so, will be in error by an amount depending on the isobaric surface selected. The actual dynamic relief of the isobaric surface will be one among the infinite number of possible reliefs that can be drawn within the error limits. In this section, the selection of a dynamic relief is done by a method of smoothening suggested by Fomin (1964) which will provide a more dependable picture of the dynamic topography of the isobaric surface as well as the vertical sistribution of geostrophic current.

The dynamic height anomalies of the different isobaric surfaces are computed, as explained in section 7.2, for the stations which lie approximately along 150 latitude in the Arabian sea. The computed dynamic reliefs of the different isobaric surfaces are then drawn and shown with dashed lines in Figs. 5-8. The lines on both sides of the computed dynamic profile are the limits of the error interval. The error limits taken were  $1/\sqrt{2}$  times the maximum error for the concerned isobaric surface because the dynamic relief of an isobaric surface in a profile highlights the differences in dynamic height anomalies between adjacent stations and, as we have seen earlier (section 5.2), the total error in the difference in dynamic height anomalies is not the sum of the errors but the sum divided by  $\sqrt{2}^*$ . The actual dynamic relief of an isobaric surface will be one among the many that are possible between the error limits. Those that are selected for the different isobaric surfaces are shown by thick lines in the figures and are the smoothest possible curves between the error limits. The smoothest

<sup>\*</sup>Fomin (1934) took the error limits as equal to the maximum error itself which, in the light of the above, should be considered as incorrect.

-124-


Fig. 5: Dynamic relief of different isobaric surfaces along approximately 15°N Latitude in the Arabian Sea.



approximately 15 N Latitude in the Arabian Sea.



Fig. 7: Dynamic relief of different isobaric surfaces along approximately  $15^{\circ}$  N Latitude in the Arabian Sea.



Fig. 8: Dynamic relief of different isobaric surfaces along approximately 15° Latitude in the Arabian Sea.

possible curve is selected because under oceanic condition, in general, large horizontal gradient of any parameter should be considered as unnatural and because any distortion to the curve between the error limits should be considered as unjustified<sup>\*</sup>.

From the smoothened curves, obtained as above, the new values of the dynamic height anomalies of different isobaric surfaces for all the stations in the profile are obtained.

Smoothening is a process very commonly employed in Physical Oceanography in the processing and analysis of oceanographic data. For example, Montgomery (1954) and Stroup (1954) recommended a method of analysis of serial oceanographic data in which the shape of the station curves for any particular station is influenced by the data from the mearby stations and this procedure, they claimed, ensure merichon al continuity of features of distribution. This is nothing but smoothening of data in-between stations. Again, oceanographers generally 'follow the trend' when they draw isolines of any parameter in sections and are not very particular to draw the lines through the exact plotted points.

### CHAPTER VIII

#### COMPUTATION OF GEOSTROPHIC CURRENTS

We have obtained the smoothened values of the dynamic height anomalies along the profile with reference to 1500 d bar surface for the stations located approximately along  $15^{\circ}$ N latitude in the Arabian sea (Fig.3) in section 7.3. These smoothened values are used for the computation of relative currents using the Helland-Hansen formula. The values thus obtained are to be reduced to 3000 d bar surface to obtain geostrophic currents with reference to the selected zero reference surface (section 6.2). This chapter deals with the procedure for the reduction of the computed geostrophic currents to the required zero reference surface.

# 8.1. Reduction of the computed geostrophic current with reference to 1500 d bar surface to 3000 d bar surface

We have seen in chapter VI that the equation (6.10), namely:

 $V_{x}(z) = V_{x}(H) \psi (z)$ 

describes the vertical distribution of the absolute value of the geostrophic current velocity in an oceanic region where Shtokman's density model is applicable and so may be used to compute any subsurface current. It is not essential that the absolute value of the sea surface current should be known for the computation of the velocities of subsurface currents. The computation can be done if we know the difference between the geostrophic current velocities at any two depths. If we know the difference in geostrophic current velocities at the sea surface and at some depth  $z_1$ , by measurement, then the absolute value of the geostrophic current velocity at the sea surface,  $V_{\chi}(H)$ , can be obtained by sub-tracting  $V_{\chi}(H)$  from both sides of equation (6.10) and then solving for  $V_{\chi}(H)$ .

$$V_{x}(z_{1}) - V_{x}(H) = V_{x}(H) \phi (z_{1}) - V_{x}(H)$$

$$= V_{x}(H) [\phi (z_{1}) - 1]$$

$$\therefore V_{x}(H) = \frac{V_{x}(z_{1}) - V_{x}(H)}{\phi (z_{1}) - 1}$$

$$= \frac{V_{x}(H) - V_{x}(z_{1})}{1 - \phi (z_{1})}$$
(8.1)

Substituting equation (8.1) in equation (6.10), a formula for computing the vertical distribution of geostrophic current velocity from a knowledge of the relative value

-127-

of geostrophic current between two depths is obtained as

$$V_{x}(z) = \frac{[V_{x}(H) - V_{x}(z_{1})] \phi(z)}{1 - \phi(z)}$$
(8.2)

We have already seen in section (6.2) that Shtokman's density model may be applied to the oceanic mass field below 1400 m. Taking the term  $[V_x(H) - V_x(z_1)]$ of equation (8.2) as the relative current at 1400 d bar surface relative to 1500 d bar surface. It can be computed using the Helland-Hansen method. Also, the stratification function  $\Phi$  (z) relative to the assumed zero surface at 3000 m depth can be computed for the same region. Once the quantities  $[(V_x(H) - V_x(z_1))]$  and  $\phi(z)$  are computed, the geostrophic current relative to 3000 d bar surface at any depth below 1400 d bar surface can be obtained using equation (8.2). The computed currents, relative to . 1500 d bar surface, at different isobaric surfaces above 1400 d bar surface, can be reduced to the values relative to 3000 d bar surface by adding algibraically to these computed values the currents computed at 1500 d bar surface. The current at 1400 d bar surface may be obtained either by adding the current computed at 1500 d bar surface or by the use of equation (8.1).

The method of computation of stratification function  $\phi$  (z) for station No.82 is shown in Table IX, assuming that the lower boundary of the current is 3000 m. For the purpose of this computation, it is convenient to write the equation for the stratification function  $\phi$  (z) in the finite difference form as

$$\psi(z) = \frac{\int_{0}^{z} \delta(z) dz}{\int_{0}^{z} \delta(z) dz} = \frac{\tilde{\delta}(z) \Delta z}{\int_{0}^{z} \delta(z) dz} = \frac{\tilde{\delta}(z) \Delta z}{\tilde{\delta}(z) \Delta z}$$
(3.3)

The first six steps of the computation of function  $\phi$  (z) have already been explained in section 6.2. The seventh column of Table IX gives the value of  $\phi$  (z) as obtained by the division of the different values in the sixth column by the value at the top of the column. Similarly the function  $\phi$  (z) was computed for the stations 48,83,103 and 111 and are tabulated in Table X. The stations used are the same as those used for the computation of the function F(H) and are well spread over the whole of the Arabian sea (Fig.3). Table X shows that the function  $\phi$  (z) is not very different for these stations which means that in the region selected water is very uniform in the horizontal plane as already seen from Table VIII. An average value of the function  $\phi$  (z) is shown as the last column of Table X. The deviation of the function  $\phi$  (z) at any station from this average value will not be greater than 4 units in its second decimal place. This

## Table IX

							_
H-z (m)	σ <sub>t</sub>	K <b>ð</b> (z)	к ð(z)	κ <b>δ</b> (z) 4	$\Delta z K \sum_{0}^{z} \delta(z) \Delta z$	φ (z)	-
1	. 403 (707 (223 (726 (446 (467 (72 (72 (72 (72 (72 (72 (72 (72 (72 (7	n na ca ra na ca na ca ca ca ca	4 	5	6	7	
1400	27,65	0,16	0.14	14.0	64.0	1200	
1500	27.69	0,12	0.09	22.5	50.0	0.78	
1750	27 <b>.</b> 75	0.06	0.05	12.5	27.5	0.43	
2000	27.78	0.03	0.02	10.0	15.0	0.23	
2500	27.80	0.01	0.01	5.0	5.0	0.08	
3000	27.81	0.00			0.0	0.00	_

Computation of stratification function  $\phi$  (z)

### Table X

Values of  $\varphi$  (z) for different stations for an assumed reference surface at 3000 m

 Н-z	an ann ann ann a' a' a' a' ann ann a'	Station	n numbers			
(m)	48	82	88	103	111	ψ (2)
1400	1.00	1.00	1.00	1.00	1.00	1.00
1500	0.81	0.78	0.82	0.80	0.80	0.81
1750	0.47	0.43	0.49	0.47	0.47	0.47
2000	0.27	0.23	0.26	0.27	0.27	0.26
2500	0.07	0.08	0.07	0.07	0.07	0.07
3000	0.00	0.00	0.00	0.00	0.00	0.00

means that, in view of equation (6.10), the error in the computed current using the average value of the function  $\phi$  (z) will not be greater than 4/ of its value relative to 3000 d bar surface.

The velocity at z = H given by equation (8.1), namely,

$$V_{x}(z) = \frac{[V_{x}(H) - V_{x}(z_{1})]}{1 - \phi(z_{1})}$$

will be correct only if the absolute value of the relative current  $[V_x(H) - V_x(z_1)]$  is known by measurement. If the relative current is obtained by computations using classical Helland-Hansen method, an error will be introduced into the computed value of the geostrophic velocity at z = H due to the error committed in the computation of the density of sea water. The magnitude of this error may be estimated as follows: Using formula (5.18) the relative current between two depths, z = H and  $z = z_1$ , is obtained in the classical Helland-Hansen method as:

$$V_{x}(H) - V_{x}(z_{1}) = \frac{10(\Delta D_{A} - \Delta D_{B})}{L \cdot 2 \omega \sin \phi}$$
(3.4)

If this value of the relative current is substituted in equation (8.1), we get

$$V_{x}(H) = \frac{10(\Delta D_{A} - \Delta D_{B})}{L \cdot 2 \omega \sin \phi [1 - \phi (z_{1})]}$$
$$= \frac{10(\Delta D_{A} - \Delta D_{B}) \int_{0}^{H} \delta(z) dz}{L \cdot 2\omega \sin \phi [\int_{0}^{H} \delta(z) dz - \int_{0}^{z_{1}} \delta(z) dz]}$$
(8.5)

By taking the total differential of equation (8.5) and dividing by the equation, we get the relative error in the computation of the absolute current velocity at z = H as:  $\frac{dV_x(H)}{V_x(H)} = \frac{d(\Delta D_A - \Delta D_B)}{\Delta D_A - \Delta D_B} + \frac{2d\sigma_t [z_1 \int_{H}^{H} \delta(z)dz - H \int_{I}^{Z_1} \delta(z)dz]}{\int_{0}^{H} \delta(z)dz \cdot \int_{1}^{H} \delta(z)dz}$ (8.6)

The magnitude of the relative error in the determination of the geostrophic current at 1400 m relative to 3000 m by the application of Shtokman's density model if the relative current at 1400 m relative to 1500 m is obtained by Helland-Hansen method may be computed using the expression (8.6). The first term on the right hand side of this expression is approximately equal to unity because the error in the difference in dynamic height anomaly at 1400 d bar surface relative to 1500 d bar surface will be of the same order of magnitude as the difference in dynamic height anomalies itself. The

-131-

contribution of the second term on the right hand side of expression (8.6) may be estimated using Table IX to get the values of the different integrals. This value is about 0.7. Hence the total relative error is 1.7 or 170%. But if the classical method is used for the computation of current at 1400 d bar surface relative to 3000 d bar surface, without the application of Shtokman's model, the relative error may be 500% or more. It may be expected that the error from this source becomes negligibly small when the dynamic relief of the 1400 d bar surface is smoothened as explained earlier in section 7.3.

Using the smoothened values of the dynamic height anomalies of the different isobaric surfaces for the stations in the profile approximately along  $15^{\circ}$  A latitude in the Arabian sea (Fig.3), obtained using the methods explained earlier, the relative currents relative to 1500 d bar surface were computed at all the different isobaric surfaces using classical Helland-Hansen method. Then using the average value of the function  $\phi$  (z) given in Table X, the geostrophic currents at the different depths below 1400 d bar surface relative to 3000 d bar surface were computed. Using these computed currents at 1500 d bar surface, the computed currents above this level were adjusted to reduce them to values relative to

-132-

3000 d bar surface as explained earlier. The results are used to draw the vertical distribution of geostrophic current velocity along the profile relative to 3000 d bar surface and are shown as Figs. 9 and 10. Figs. 11 and 12 are the corresponding vertical distribution of geostrophic current velocities when the computations were done using classical methods.

Comparison of Figs. 9 and 10 with the corresponding Figs. 11 and 12, brings out certain important observations. Results of the computation of geostrophic current using smoothened values of the dynamic height anomalies of the different isobaric surfaces show a substantial northerly current between stations 41 and 42. The classical method yields only an erratic current of small magnitude in this region and this current changes direction as the depth increases. The fact that a substantial northerly current does exist in the region is evident from the raising trend of the isanosteres between the stations towards the coast shown in the diagram depicting the vertical distribution of thermosteric anomaly along the same profile produced by Sastry and D'Souza (1971) and reproduced as Fig.13 in this work. The failure of the classical method here may be due to the high computational errors arising out of the small distance between the stations.

-133-

### Figs. 9 and 10:

Geostrophic current (cm/sec) relative to 3000 decibars along approximately  $15^{\circ}$ N Latitude in the Arabian Sea, computed using the smoothened data (flow is towards south in the shaded area).





### Figs. 11 and 12:

Geostrophic current (cm/sec) relative to 3000 decibars along approximately 15<sup>°</sup>N Latitude in the Arabian Sea, computed using classical methods (flow is towards south in the shaded area).



DEPTH IN METRES





Fig. 13: Thermosteric anomaly (al/T) along approximately 15<sup>°</sup>N Lat. in the Arabian Sea (after Sastry and D'Souza,1971)

Another observation is that the computed currents which are within the limits of accuracy of computation, gets smoothened out in the smoothening process. Under certain situations the smoothening procedure causes a change in the direction of such currents. Thus the southerly currents below 200 m between stations 44 and 45, below 1600 m between stations 55 and 57 and above 100 m between stations 48 and 49 seen in Figs.11 and 12 are seen to be northerly in Figs.9 and 10. Similarly the northerly currents below 300 m between stations 50 and 51 and below 1200 m between stations 49 and 50 in Figs.11 and 12 are seen to be southerly in Figs.9 and 10. The magnitudes of computed currents is generally decreased by the smoothening but under certain situations an increase can also occur as we have already seen between stations 41 and 42 where the computational error is very large.

### 8.2. Summary and conclusions

The measurement of the independent variables at sea is subject to random errors. These errors get propagated, through the series of computational procedures, into the final derived results. The magnitude of errors in the final results may, at times, be of such magnitude as to vitiate the results themselves. The present study

-134-

is an attempt to examine the limits of errors contained in the basic data and the magnitude of the error component in the derived results of Dynamical Oceanography.

The general theory of errors relevant in the context of the studies on the propogation of errors in physical oceanographic computations have been discussed in chapter II. After discussing the type of errors in section 2.1, the general formula for errors is derived in section 2.2. The applications of the general formula for errors to the fundamental operations of arithmetic have been dealt with in section 2.3. In section 2.4 after discussing the normal law of errors, an expression for the computation of the standard deviation,  $\sigma$  of a normal error distribution representing the distribution of errors in the summation of S values, when only the maximum individual error is known, is derived.

The errors in the measurements of the independent oceanographic variables, namely, temperature, pressure and salinity are discussed in chapter III. In section 3.1 is discussed the magnitude of error committed in the measurement of temperature using the protected reversing thermometer. There it is shown that addition of expansion correction increases the magnitude of error in the temperature measurement generally and that in a tropical

-135-

sea the magnitude of the total error in the temperature measurement should be taken as  $\pm 0.01^{\circ}$ C upto a depth of 200 m and as  $+0.02^{\circ}$ C below the depth of 200 m. The error in the determination of sea pressure is dealt with in section 3.2, It is shown that the error committed in the replacement of pressure expressed in decibars by depth expressed in metres is negligibly small in the tropical seas. The error committed in the reading of an unprotected reversing thermometer is also discussed in this section. It is seen that addition of expansion correction increases the magnitude of error in the reading. The magnitude of the total error in this case also is equal to  $+0.01^{\circ}$ C up to a depth of 200 m and  $+0.02^{\circ}$ C below the depth of 200 m. The corresponding errors in the depth measurements are equal to  $\pm 2m$  upto a depth of 200 m and +4 m below the depth of 200 m. The errors committed in the determination of salinity are discussed in section 3.3. The error committed in the determination of salinity by the method of titration is  $\pm 0.04\%$ . and by the method of conductivity ratio measurement is +0.02%.. The accuracy of measurements using in situ instruments is discussed in section 3.4. It is shown that the accuracy obtained is much poorer compared to the classical method of obtaining the values of the independent variables.

-136-

Errors in the computation of dependent oceanograph variables due to the propogation of errors are discussed in chapter IV. In section 4.1 it is seen that irrespecti of the formula employed for the computation of  $\sigma_t$ , the error in the density determination shoul be taken as  $\pm 0.03$  in  $\sigma_t$  when salinity is determined using the method of titration and as  $\pm 0.02$  in  $\sigma_t$  when salinity is determined using the method of conductivity ratio measurement. The corresponding errors in the computation of specific volume enomaly is shown to be equal to  $\pm 3$  cl/ and  $\pm 2$  cl/ton respectively in section 4.2.

In chapter V the errors committed in the computations of derived quantities, namely, dynamic heigh anomaly of isobaric surfaces and relative currents are discussed. In section 5.1 the magnitudes of errors in th computation of dynamic height anomaly of different isobaric surfaces relative to a reference isobaric surface at 1500 m are obtained using the normal law of It is found that the deeper the reference surfac errors. selected, the larger is the error in the determination of the dynamic height anomaly of an isobaric surface. The of magnitudes of errors in the computation/relative currents is discussed in sectior 5.2. It is seen that the magnitude of error in the computation/relative current is

-137-

directly proportional to the error in the determination of the difference in dynamic height anomalies between two stations and is inversely proportional both to the distance between the two stations and the sine of the latitude angle. It is also seen that if the distance between stations is 100 km, the absolute magnitude of error in the determination of relative current with reference to 1500 d bar surface is less than 1 cm/sec above 40° latitude angle and increases towards the equator. Very near the equator the rate of increase is very large.

After critically reviewing the existing methods for the identification of a level of no motion in section 6.1, in section 6.2 a method for the identification of a level where the geostrophic current is negligibly small, by the application of Shtokman's density model to the oceanic massfield is discussed. It is shown that Shtokman's density model, may be applied to the mass field of the ocean below roughly 1500 m. A function F(H)is computed in this section by the application of the model to the oceanic mass field below 1400 m. This function decreases in magnitude as the depth of the selected reference surface increases and becomes constant at a depth equal to 3000 m. This depth, below which the function F(H) assumes a constant value, is shown, to be

-138-

a level where the magnitude of the geostrophic current is negligibly small.

In section 7.1, the existing methods for the extension of the selected zero reference surface into shallow regions for the purpose of computation are In section 7.2 a new method is critically examined. proposed for the same purpose which essentially consists of the extrapolation of the dynamic height anomaly in a profile of the isobaric surface which represents the deepest sompling depth at the shallow station. It is shown that this method is free of the objections raised in connection with the existing methods. A method of smoothening of the dynamic relief of an isobaric surface in a profile within certain error limits, is described in section 7.3. It is shown that the error limit for the above purpose should be  $1/\sqrt{2}$  times the maximum error computed for the dynamic height anomaly of the isobaric surface concerned. It is also shown that the smoothened values of the dynamic height anomalies of isobaric surfaces will give a more dependable picture of the oceanic circulation pattern.

In sec. 8.1 of chapter VIII, the reduction of the computed current relative to 1500 d bar surface to

-139-

the zero reference surface at 3000 m depth is dealt with. It is shown that the relative current computed at 1400 d bar surface relative to 1500 d bar surface may be used for obtaining the currents at different isobaric surfaces below 1400 m, if the stratification function,  $\phi$  (z), resulting from the application of Shtokman's density model to the region, is known. The current computed at different depths above 1500 d bar surface may be obtained by the algebraic addition of the current at 1500 d bar surface to the different computed currents. The results thus obtained for the stations lying approximately along 15°N latitude in the Arabian sea are compared with results obtained using the classical method.

The studies showed that the graphical method suggested by Montgomery (1954) and Montgomery and Wooster (1954) for the determination of thermosteric anomaly,  $\delta_{\rm T}$ , introduces, in the computed current, an error, that is equal to 2 times the error committed when the values are obtained using classical methods. Consequently, when the values of  $\delta_{\rm T}$  are obtained graphically, the computed circulation pattern may become undependable, particularly when the circulation is weak and is in the equatorial latitudes.

-140-

Proper selection of a zero reference surface is important for the presentation of an oceanic circulation pattern. If a reference level is selected arbitrarily and if a current does exist at that level, the computed currents in the layers above will be different from the actual currents. Also the wrong selection of a reference level will result in the computation of an oppositely directed current below this level which increases in magnitude towards the bottom, a result which is against the present day oceanographic experience.

Since the error in the computed current increases very rapidly towards the equator, computed current using the classical method is very undependable in these regions, particularly when the current is weak. Smoothening of the dynamic relief, which may be successful in the higher latitudes for the presentation of a dependable circulation pattern, may prove unsuccessful very near the equator since small variations in the values of dynamic height anomalies that produce large currents in these latitudes will be smoothened out, if they are within the error limits.

It is seen that decrease in station spacing will increase the error in the computed current and may lead to undependable results, particularly, when the values of

-141-

thermosteric anomalies are obtained using the graphical method suggested by Montgomery (1954) and Montgomery and wooster (1954). The error in the computed current committed when this method is used, is again doubled if the station spacing is halved. This means that the error in the computed current will be 4 times that given in Fig.2 if we take the station spacing as equal to 50 km. Hence the error in the computed current when the graphical method is used to obtain the required data and when the station spacing is 50 km will be +4 cm/sec above 40° latitude, and +36 cm/sec at 5° latitude. Such large errors may induce a band structure in the computed current pattern, particularly in the lower latitudes, when station spacing is reduced, a phenomenon observed by Montgomery and Stroup (1962). Hence it is advis able not to go for decrease in station spacing, in the hope of getting a detailed current structure. The station spacing, preferably, should not be less than 1° latitude or longitude angle.

#### REFERENCES

- \*Bjerknes, V. (1898): <u>Uber Cinen hydrodynamischen</u> <u>Fundamentalsatz und seine Anwardung besonders auf</u> <u>die Mekanik der Atmosphare und des Weltmures</u>., <u>K. Svenska Vet. Acad. Hand</u>., 31(4).
- \*Bjerknes, V. (1900): Das dynamische Prinzip.der zirkulationsbewegung in der Atmosphare. Meteorol., Z.
- Bjerknes, V. and Sandstrom, J.W. (1910): Dynamic Meteorology and Hydrography. Part I., Statics, Carnegie Inst., Washington DC., Publ. No. 88, 146 pp.
- Bradshaw, A. and Scheliecher, K.E. (1965): The effect of pressure on the electrical conductance of sea water, Deep Sea Res., 12: 151.
- Brown, N.L. and Hammon, B.V. (1961): An inductive salinometer, <u>Deep Sea Res.</u>, 8: 65.
- \*Bumpus, D.F. and Mertineau, D. (1948): Extension of Ennis's table for the determination of density (σ<sub>t</sub>), <u>Woods Hole Oceanographic Institution Tech. Rep</u>.
- Callaway, E.B. (1950): <u>Graphical Determination of Specific</u> <u>Volume Anomalies and Current</u>, U.S. Navy Hydrographic Office, Washington, D.C.
- Cox, R.A., Culkin, F. and Riley, J.P. (1967): The electrical conductivity/chlorinity relationship in natural sea water, <u>Deep Sea Res.</u>, 14: 203.
- Cox, R.A., McCartney and Culkin, F. (1968): Pure water for relative density standard, <u>Deep Sea Res</u>., 15: 319.

- Cox, R.A., McCartney and Culkin, F. (1970): The specific gravity/salinity/temperature relationship in natural sea water, Deep Sea Res., 17: 679.
- Culkin, F. (1965):in Riley, J.P. and Skirrow, G. (ed.) <u>Chemical Oceanography</u>, Vol. I, Accussic Press, New York, pp. 122-162.
- Defant, A. (1941a): Die absolute Berechnung Ozeanischer Strome nach dem dynamischen Verfahren., Ann. Hydrograph. Marit. Meteorol., No.6.
- Defant, A. (1941b): <u>Die absolute Topographie des</u> <u>physikalischen Meeresniveaus und der Druckflachen,</u> <u>sowie die Warserbewegungen im Atlantischen Ocean,</u> <u>Deutsche Atlantische Exped. 'Meteor' 1925-1927,</u> 6(5): 191.
- Dietrich, G. (1936): <u>Aufbau und Bewegung von Golfstrom und</u> Aqulhasstrom, <u>Naturwissenschaften</u>, No.15.
- Dobrovol'skii, A.D. (1949): Contribution to the problem of the location of the zero surface for dynamic computation in the North Pacific, <u>Tr. Inst. Okeanol.</u> <u>Akad. Nauk SSSR</u>, Vol.4.
- Dittmar, W. (1884): in <u>The voyage of H.M.S. Challenger</u> Murray, J. (ed.) Vol.I, H.M. Stationery Office, London, pp. 1-251.
- Ekman, V.W. (1908): <u>Die Zusammendruck barkeit des</u> <u>Murwassers. Conseil. Perm. Intern. p. 1'Explor,</u> <u>de la Mer</u>, <u>Pub. de Circonstance</u>, No.43, 47 pp.

- Ennis, C.C. (1944): Note on the computation of the density of sea water and on the correction of deep-sea reversing thermometers, <u>Carnegie Inst. Wash</u>. Pub. No.545, pp. 23-42,
- Fleming, R.H. (1939): <u>Tables for sigma-T</u>, <u>Jour. Mar. Res.</u>, 2: 9-11.
- Fomin, L.M. (1964): <u>The Dynamic Method in Oceanography</u>, Elsevier Publishing Company, New York, 212 pp.
- Forch, C., Knudsen, M. and Sorensen, S.P.L. (1902): Kgl. Dan. Vidensk. Selsk. Raekke, naturvideusk, og methem Afd, XII, I, Skrifter, 6, 151 pp.
- Goren, P. (1948): Methods for estimating dynamic slopes and currents in shallow water. J. Mar. Res., 6(3).
- Goulet, J.R. and Culverhouse, B.J. (1972): J. <u>Geophys. Res</u>., 77: 4588.
- Ĝrasshoff, K. (1968): ICES, CM 1968/C, 39, Hydr. Comm.
- Grasshoff, K. (1976): in Grasshoff, K. (ed.) <u>Methods of</u> <u>Sea water Analysis</u>, <u>Verlag</u>, <u>Chemie</u>. <u>Weinheim</u>, New York, p. 31.
- Ĝrasshoff, K. and Hermann, F. (1975): I.C.E.F., CM 1975/C, 46, Hydr. Comm.
- Helland-Hansen, B. (1916): <u>Nogen Hydrografiske metoder</u>. <u>Skand. Naturforsker mote</u>, Kristiania (Oslo), p. 357.
- Helland-Hansen, 3. (1934): The Sognefjond section, James Johnstone Memorial Volume, Liverpool Uni. Press, pp. 257-274.
- Hidaka, K. (1940a): Absolute evaluation of ocean currents in dynamical calculations, <u>Proc. Imp. Acad.</u>, Tokyo, 16(8): 391.

- Hidaka, K. (1940b): A practical evaluation of ocean currents, <u>Proc. Imp. Acad.</u>, lokyo, Vol.16.
- Hidaka, K. (1949): Depth of motionless layer as inferred from the distribution of salinity in the oceans, Trans. Am. Geophys. Union, 30(3).
- Hidaka, K. (1950): An attempt to determine the motionless layer; in the ocean, <u>Oceanogr. Mag.</u>, 2(1).
- Iselin, C. O'D. (1936): A study of the circulation of the Western North Atlantic, Papers Phys. Oceanog. and Meteor., 4(4): 101.
- Jacobsen, J.P. (1916): Contribution to the hydrography of the Atlantic, <u>Medd. Komm. Havundersgelser</u>, <u>Ser.</u> <u>Hydrografi</u>, Vol.2.
- Jacobsen, J.P. and Jensen, A.J. (1926): Examination of hydrological measurements from the research vessels 'Explorer' and 'Dana' during the summer of 1924, Rapp. Conseil Perm Intern. Exploration Mer., Vol.39.
- Jacobsen, J.P. and Knudsen, M. (1940): Urnormal 1937 or primary standard sea water 1937, <u>Internat. Assn.</u> <u>Phys. Oceanogr. Pub. Sci.</u>, 7, 38 pp.
- Jakhelln, A. (1936): The water transport of the gradient currents, <u>Geofys</u>. Publikajoner, 11(11): 1.
- Knudsen, M. (1901): <u>Hydrographical Tables</u>, G.E.C. Gad, Copenhagen, 63 pp.
- Knudsen, M. (1903): Pub. Circ. Cons. Perm. Int. Explor. Mer. 5, pp 11-13.

- Knudsen, M., Forch, C. and Sorensen, S.P.L. (1902): in <u>Wissenschaft liche Meeresumtersachungen</u> (<u>Neve Folge</u>), <u>Bd.6</u>, <u>Lipsias und Tischer</u>, Kiel, pp. 123-184.
- Kratky, O., Leopold, H. and Stablenger, H. (1969): Dichtemessungen an Flussigkeiten und Gasen auf 10<sup>-5</sup>c/cm bie O.6 cm proparatvolumen, Z. Angew. Phys., 27: 273.
- Kremling, K. (1971): Measurement of sea water density by a new laboratory method, <u>Nature</u>, 229: 109.
- Kremling, K. (1972): Comparison of specific gravity in natural sea water from hydrographical tables and measurements by a new density instrument, <u>Deep Sea Res.</u>, 19: 377.
- \*LaFond, E.C. (1940): <u>Practical Oceanographic Tables</u>, Univ. of Calif., Scripps Institution of Oceanography
- LaFond, E.C. (1951): <u>Processing Oceanographic Data</u>, H.O. Pub.No.614, U.S. Navy Hydrographic Office, Washington, D.C., 114 pp.
- Layman, J. (1969): Redefinition of salinity and chlorinity, Limnol. Oceanogr., 14: 928.
- Layman, J. and Fleming, R.H. (1940): Composition of sea water, J. Mar. Res., 3: 134.
- \*Mamaev, O.L. (1955): Methods of determining the zero dynamic surface in the world ocean, Vestn. Mosk. Univ., No.10.
- Mathews, D.J. (1932): <u>Tables for the Determination of the</u> <u>Density of the Sea Jater under Normal Pressure</u>, <u>Conseil Perm. Intern. P. l'Explor. de la Mer</u>, Copenhagen, 59 pp.
- McEwen, G.H. (1929): Tables to Facilitate Dynamic Computations of Ocean Currents according to the Bjerknes Circulation Theory, Univ. of Calif., Scripps Institution of Oceanography.
- Montgomery, R.B. (1954): Analysis of a <u>Hugh M. Smith</u> a ceanographic section from Honolulu gouthward across the equator, J. Mar. Res., 13: 67.
- Montgomery, R.B. and Stroup, E.D. (1962): Equatorial Waters and Currents at 150° 1 in July-August 1952, John Hopkins Oceanogr. Stud., No.1, 68 pp.
- Montgomery, R.B. and Wooster, W.S. (1954): Thermosteric anomaly and the analysis of serial oceanographic data, <u>Deep Sea Res.</u>, 2: 63.
- Morris, A.W. and Riley, J.P. (1964): Direct gravimetric determination of salinity of sea water, <u>Deep Sea Res.</u>, 11: 899.
- \*Murray, J. (1893): Trans. Roy. Soc., Edinborough, 37: 481.
- \*Park, K. (1965): J. Oceanogr. Soc. Japan, 21: 124.
- \*Parr, A. (1938): Analysis of current profile by a study of pycnomeric distortion and identifying properties, <u>J. Mar. Res.</u>, No.4.
- \*Perkin, R.G. and Walker, E.R. (1972): <u>J. Geophys. Res</u>., 12: 6618.

- Riley, J.P. (1965): in Riley J.P. and Skirrow, G. (ed.) <u>Chemical Oceanography</u>, Vol.3, Academic Press, London, p. 295.
- Riley, J.P. (1975): in Riley, J.P. and Skirrøw, G. (ed.) <u>Chemical Oceanography</u>, Vol.3, (2nd Edition), Academic Press, London, p. 193.
- Rossby, C.G. (1936): Dynamics of steady ocean currents in the light of experimental fluid gynamics, <u>Papers Phys. Oceanog. and Meteor.</u>, 5(1) (Cambridge, Mass.)
- Scarborough, B. (1966): <u>Numerical Mathematical Analysis</u> (6th Edition), The Johns dopkins Press, Baltimore, U.S.A., 600 pp.
- Sandstrom, J.W. and Helland-Hansen, B. (1903): <u>Uber die</u> <u>Berechung von Meeresstromungen, Rep. Norweg. Fish.</u> <u>Mar. Invest.</u>, 2(4): 1.
- Sastry, J.S. (1971): Proc. Symp. Indian Ocean and Adjacent Seas - Their Origin, Science and Resources, Cochin (India).
- Sastry, J.S. and D'Souza, R.S. (1971): Oceanography of the Arabian sea during the southwest monsoon season -Part II: Stratification and circulation, <u>Indian J. Met. Geophys.</u>, 22: 23.
- Sastry, J.S. and D'Souza, R.S. (1972): Oceanography of the Arabian sea during the southwest monsoon season -Part III: Salinity, Indian J. Met. Geophys., 23(4): 479.
- Seiwell, H.R. (1937): The minimum oxygen concentration in the western basin of the North Atlantic, <u>Papers Phys. Oceanog. and Meteor.</u>, 5(3).

- Shtokman, V.B. (1950): Determination of current velocities and of density distribution in a transverse section of an infinite channel as a function of the wind effect and lateral friction, <u>Dokl. Akad. Nauk SSSR</u>, 71(1).
- \*Shtokman, V.B. (1951): Determination of stationary currents and of the field of mass resulting from wind in a baroclinic sea, <u>Tr. Inst. Okeanol. Akad. Nauk SSSR</u>, Vol.6.
- Shumacher, A. (1923): <u>Neue Hilfstafeln fur die</u> <u>Unikippthermometer nach Richter und Bietrage zur</u> <u>thermometrischen fiefenmessung</u>, <u>Ann. Hydr. Marit. Met.</u>, 5(51): 271.
- Sorensen, S.P.L. (1902): in Knudsen M. (ed.) <u>Berichte uber</u> die Konstantenbestimmungen zur Aufstellung der <u>hydrographischen Tabellen</u>, 3, Copenhagen, pp. 93-138.
- Stroup, E.D. (1954): Mid-Pacific oceanography, Part IV, transequatorial waters January-March 1952, Spec. Sci. Rep. U.S. Fish. Wildl. Serv., Fish No.135, p. 52.
- Sund, O. (1926): Graphical calculation of specific volume and dynamic depth, Jour. Conseil, 1: 233.
- Sverdrup, H.U. (1933): <u>Verinfachtes Verfahren zum</u> <u>Berechnung der Druck und Massenverteilung im Meere,</u> <u>Geofys</u>, Pub. 10(1): 1.
- Sverdrup, H.U. (1938): On the explanation of the oxygen
  maxima and minima in the oceans, Conseil. Perm.
  Intern. P. 1'Expl. de la Mer, J. Cons., 13(2): 163.

- Sverdrup, H.U., Johnson, M.W. and Fleming, R.H. (1942): <u>The Oceans - Their Physics, Chemistry and General</u> <u>Biology</u>, Prentice Hall Inc., Englewood Cliffs, N.J., 1027 pp.
- Thompson, E.F. (1939): A rapid method for the determination of dynamic heights (or depths) at successive lowerings at an anchor station, J. Mar. Res., 2(2).
- UNESCO (1962): Tech. Papers in Mar. Sci., No.1, 29 pp.
- UNELCO (1965): Tech. Papers in Mar. Sci., No.4, 29 pp.
- UNESCO (1966): <u>International Oceanographic Tables</u>, National Institute of Oceanography of Great Britain and UNESCO, 118 pp.
- UNESCO (1968): International Oceanographic Tables, National Institute of Oceanography of Great Britain and UNESCO, 128 pp.
- \*Walker, E.R. and Chapman, K.D. (1973): <u>Pacific Marine</u> <u>Science Report</u>, 73-5, 52 pp. (Unpublished manuscript Pacific Marine Sciences Directorate, Victoria B.C.
- Wattenberg, H. (1938): Die Vertielung des Sanerstoffs im Atlantischen Ozean. Meteor - Werk, 9(1) (Berlin).
- Wilson, T.R.S. (1975): in Riley J.P. and Skirrow, G. (ed.) Chemical Oceanography, Vol.I (2nd edition), Academic Press, London, pp. 365-413.
- Wooster, W.S., Lee, A.J. and Dietrich, G. (1969): Redefinition of salinity, <u>Deep Sea Res</u>., 16: 321.





- \*Wust, G. (1935): <u>Schichtung und Zirkulation des Atlantischen</u> Ozeans, Die Stratosphare, Wiss Ergeb. Deut Atlant. Exped. 'Meteor' 1925-1927, Vol.6, Part 2.
- Wust, G. (1955): Stromgeschwindigkeiten im tiefen und boden - Wasser auf Grund dynamischer Meteor -Profile der Duetschen Atlantischen Expedition 1925-1927, Deep Sea Res., Suppl. to Vol.3.
- Young, H.D. (1962): Statistical Ireatment of Experimental Data, McGraw Hill Book Co. Inc., New York, 172 pp.

Not referred to in original.